DISTRIBUTED OUTPUT FEEDBACK INDIRECT MRAC OF CONTINUOUS-TIME MULTIAGENT LINEAR SYSTEMS*

JIAN GUO[†], YANJUN ZHANG[‡], AND JI-FENG ZHANG[§]

Abstract. This paper studies the distributed leader-follower output consensus problem for continuous-time uncertain multiagent linear systems with general input-output forms. Specifically, we extend the well-known output feedback indirect model reference adaptive control (MRAC) and develop a fully distributed output feedback indirect MRAC scheme to achieve closed-loop stability and asymptotic leader-follower output consensus. Compared with the existing results, the proposed distributed MRAC scheme has the following characteristics. First, the orders of each agent's pole/zero polynomials, including the followers and the leader, can differ from others, and the parameters in each follower's pole/zero polynomials are unknown. Second, the proposed adaptive control law of each follower solely relies on the local input and output information without requiring the state observer and the structural matching condition on the followers' dynamics, commonly used in the literature. Third, for any given leader with a relative degree n^* , the leader-follower output tracking error and its derivatives up to the n^* th order converge to zero asymptotically, which has never been reported in the literature. Finally, a simulation example verifies the validity of the proposed distributed MRAC scheme.

 ${\bf Key \ words.}\ {\bf model \ reference \ adaptive \ control, \ distributed \ output \ feedback, \ multiagent \ systems, \ leader-follower \ consensus$

MSC codes. 93B52, 93C15, 93C40

DOI. 10.1137/23M1620107

1. Introduction. Multiagent systems (MASs) focus on the joint behavior of autonomous agents. In the past decades, researchers in various fields focused on how agents cooperate with each other and revealed many interesting phenomena [3, 14]. A fundamental problem in MASs is designing a control law for each agent that solely relies on neighborhood information, such that the networked system can achieve specific tasks such as formation, swarming, or consensus. Several prestigious papers [4, 11] have further highlighted the important and fundamental problems the cooperative control of MASs suffers from.

Many remarkable results have been reported to deal with various multiagent distributed control and coordination tasks, e.g., consensus/synchronization [20], formation control [8, 36], bipartite consensus [18, 39], and containment control [7, 19]. Since cooperative control requires agents to reach agreement on their respective tasks, consensus control has become a central topic in MAS research. Currently, there are

^{*}Received by the editors November 27, 2023; accepted for publication (in revised form) February 7, 2025; published electronically May 9, 2025.

https://doi.org/10.1137/23M1620107

Funding: This work was supported in part by the National Natural Science Foundation of China under grants 62322304, 62173323, T2293770, and 62433020.

[†]State Key Laboratory of Mathematical Sciences, Academy of Mathematics and Systems Science, Chinese Academy of Sciences, Beijing 100190, China, and School of Mathematical Sciences, University of Chinese Academy of Sciences, Beijing 100049, China (j.guo@amss.ac.cn).

[‡]Corresponding author. School of Automation, Beijing Institute of Technology, Beijing, 100081 China (yanjun@bit.edu.cn).

[§]School of Automation and Electrical Engineering, Zhongyuan University of Technology, Zhengzhou, 450007 Henan Province, China, State Key Laboratory of Mathematical Sciences, Academy of Mathematics and Systems Science, Chinese Academy of Sciences, Beijing 100190, China, and School of Mathematical Sciences, University of Chinese Academy of Sciences, Beijing 100049, China (jif@iss.ac.cn).

mainly two consensus control strategies: the behavior-based (or leaderless) strategy [17, 24] and the leader-follower strategy [9, 43]. The main task of a consensus control problem is to design appropriate distributed consensus protocols to achieve consensus. However, designing distributed protocols is challenging due to the interaction between agents [16].

To date, the consensus problem has been extensively studied in the control community. For instance, in [24, 26], the consensus problems for some simple linear MASs were investigated. Since then, the literature has addressed the consensus control for the case with noises [51], for general linear homogeneous MASs [15, 34, 46], some nonlinear MASs, such as Lipschitz nonlinear systems [31], Euler-Lagrange systems [23], rigid body systems [27], nonlinear MASs with compasses [22] and fractional MASs [44]. Note that the well-known backstepping technique originally developed in [13] for nonlinear adaptive control design is still effective and quite popular for cooperative control design and analysis of MASs [40]. Furthermore, the output regulation technique is also a powerful tool for cooperative control design and analysis, and many remarkable results have been published [35, 41].

Adaptive control methods are widely used in various fields [42], in which the model reference adaptive control (MRAC) technique has attracted significant attention since it can simultaneously realize online parameter estimation and asymptotic tracking control for systems with large parametric/structural uncertainties [1, 10, 30, 37, 45, 48, 49]. Many key problems in cooperative control theory and applications have been well handled using MRAC-based control methods [5, 6, 21, 47, 50]. Research on distributed MRAC for open-loop reference models has been done in [25]. Moreover, [30] studied the adaptive leader-follower consensus problem for MASs with general linear dynamics and switching topologies. In [5], the authors considered that the leader's external input is not shared with any follower agent and proposed a new external input estimator in a hierarchical and cooperative manner. All these results are developed under the distributed MRAC framework.

However, how to develop a fully distributed output feedback MRAC is still an open research case. Actually, after reviewing the distributed MRAC literature, we find that the existing distributed MRAC results mainly used state feedback to solve the state consensus problems under the well-known matching condition. The latter condition requires the dynamics of the followers and the leader to meet some structural matching equations from which the ideal parameters of the nominal control laws can be calculated. The matching condition with respect to most of the real control systems is quite restrictive, and largely constrains the application range of such methods. Thus, one key technical problem that must be concerned is how to relax the restrictive matching conditions, especially for the distributed MRAC. Moreover, to the best of our knowledge, a fully distributed output feedback MRAC has never been reported yet, which faces several key technical problems to be concerned. Such problems are (i) how to estimate the unknown parameters of all followers by only using their own input and output? (ii) How to design a distributed MRAC law for each follower by only using the local input and output information? (iii) How do all leader-follower tracking errors converge to zero without persistent excitation? These technical problems have not been addressed in the literature yet. Hence, this paper systematically addresses the distributed output feedback MRAC problem and solves the above technical problems. Specifically, we develop a fully distributed output feedback MRAC scheme without requiring the restrictive matching condition. Particularly, the asymptotic convergence of the leader-follower consensus is achieved.

JIAN GUO, YANJUN ZHANG, AND JI-FENG ZHANG

Overall, this work's main contributions and novelties are as follows.

- (i) A linearly parameterized output feedback adaptive control framework is established to address the distributed leader-follower output consensus problem for linear MASs in general input-output forms. Each agent's dynamics have different pole/zero polynomials and different orders, with all coefficients being unknown.
- (ii) A fully distributed output feedback adaptive control law is developed for the considered MASs, where the adaptive control law of each follower solely relies on the local input and output information without requiring the state observer and the restrictive structural matching condition on the followers' and leader's dynamics commonly used in the literature.
- (iii) To establish the distributed output matching equation for each follower, some auxiliary systems are introduced to generate filtered signals of individual signals and neighbors' outputs. Such filtered signals are crucial to constructing the distributed matching equations from which the adaptive parameters used in the adaptive control laws can always be derived.
- (iv) The closed-loop stability and asymptotic output consensus analysis are conducted by using a gradient-based framework independent of Lyapunov functions. Particularly, the leader-follower output tracking error and its derivatives up to the n^* th order converge to zero asymptotically without persistent excitation, which has not yet been reported in the literature.

The remainder of this paper is organized as follows. Section 2 provides the problem statement and the preliminaries. Section 3 introduces the distributed output feedback MRC design and the corresponding theoretical results for providing the basic idea. Section 4 is the main part of this paper presenting the adaptive control details where the coefficients are unknown, and section 5 presents two simulation examples to illustrate our algorithm's performance. Finally, section 6 concludes this paper.

Notation. In this paper, \mathbb{R} denotes the sets of real numbers. Let s denote the differential operator, i.e., $s[x](t) = \dot{x}(t)$ with $x(t) \in \mathbb{R}^n$, $t \ge t_0$. By L^{∞} , L^2 and L^1 we denote the three signal spaces defined as $L^{\infty} = \{x(t) : ||x(\cdot)||_{\infty} < \infty\}$, $L^2 = \{x(t) : ||x(\cdot)||_2 < \infty\}$ and $L^1 = \{x(t) : ||x(\cdot)||_1 < \infty\}$ with $||x(\cdot)||_{\infty} = \sup_{t\ge t_0} ||x(t)||_{\infty}$, $||x(\cdot)||_2 = (\int_{t_0}^{\infty} ||x(t)||_2^2 dt)^{1/2}$ and $||x(\cdot)||_1 = \int_{t_0}^{\infty} ||x(t)||_1 dt$, respectively.

2. Problem statement. This section formulates the system model, the control objective, the design conditions, and the technical issues to be solved.

2.1. System model. The MAS considered in this paper is described by the following input-output form:

(2.1)
$$P_i(s)[y_i](t) = k_{pi}Z_i(s)[u_i](t), \quad t \ge 0, \ i = 1, \dots, N,$$

where N is the number of followers, $y_i(t) \in \mathbb{R}$ and $u_i(t) \in \mathbb{R}$ are the output and input of the *i*-th follower, respectively, k_{pi} is a constant referred to as the high frequency gain, and $P_i(s)$ and $Z_i(s)$ are the pole and zero polynomials with unknown coefficients, degree n_i and m_i , respectively, i.e.,

$$P_i(s) = s^{n_i} + p_{i,n_i-1}s^{n_i-1} + \dots + p_{i1}s + p_{i0},$$

$$Z_i(s) = s^{m_i} + z_{i,m_i-1}s^{m_i-1} + \dots + z_{i1}s + z_{i0}.$$

It should be noted that n_i and n_j , as well as m_i and m_j , can be different for $i \neq j$, with i, j = 1, ..., N.

The leader's output and input dynamic is

(

(2.2)
$$P_m(s)[y_0](t) = r(t),$$

1619

where $P_m(s)$ is a stable polynomial of degree n^* , and r(t) is a bounded and piecewise continuous reference input signal for the leader.

Actually, (2.2) can be chosen more general as follows: $P_m(s)[y_0](t) = Z_m(s)[r](t)$, with $Z_m(s)$ and $P_m(s)$ being two given zero and pole polynomials. But, the design and analysis for more general cases are similar to that for the case of (2.2). Therefore, for simplicity of presentation, here we choose (2.2) to conduct the distributed MRAC design and analysis. The reader can refer to [10] and [37] for more details.

Next, it is important to clarify the necessity of using the input-output form (2.1) to establish a distributed MRAC framework. Some black-box systems may not afford to build a state-space system model when no information about the internal state variables is available. However, establishing a simple input-output model without containing internal state variables is possible for such black-box systems. In this case, the input-output information is adequate for the MRAC and distributed MRAC control design and stability analysis. However, a potentially arising question is that as long as an input-output model is established, one may derive its state-space realization and still use state-space-based methods to conduct the control design and analysis. Indeed, the state-space model can be derived from the input-output model. However, from a practical viewpoint, the state-space model may sometimes be unsuitable for designing the controller because the state variables generally do not have explicit physical meanings. Therefore, addressing the cooperative control problems by using the input-output models (2.1)-(2.2) is significant.

Communication graph. Let the MAS be described by (2.1)-(2.2). The communications between these N + 1 agents are modeled as a directed graph $\mathcal{G} = \{\mathcal{V}, \mathcal{E}\}$, where $\mathcal{V} = \{v_0, \ldots, v_N\}$ is the set of nodes with v_0 representing the leader, $v_i, i = 1, \ldots, N$, representing the *i*th follower, and $\mathcal{E} \subseteq \mathcal{V} \times \mathcal{V}$ being the set of edges of \mathcal{G} . The directed edge (v_j, v_i) represents a unidirectional communication channel from agent v_j to agent v_i , i.e., agent v_i can obtain the output information from agent v_j , but not vice versa. The neighborhood of agent v_i , $i = 0, \ldots, N$, is denoted by $\mathcal{N}_i = \{v_j \in \mathcal{V} : (v_j, v_i) \in \mathcal{E}\}$. A directed sequence of the edges $(v_{i1}, v_{i2}), (v_{i2}, v_{i3}), \ldots, (v_{i,k-1}, v_{ik})$ is called a path from node v_{i1} to node v_{ik} . A directed tree is a directed graph where each node except for the root node has a single neighbor, and the root node is a source node. A spanning tree of \mathcal{G} is a directed tree whose node set is \mathcal{V} . Its edge set is a subset of \mathcal{E} . Moreover, (v_i, v_i) is called a self-loop. This study assumes a simple graph, i.e., the graph has no self-loops or multiple arcs.

2.2. Control objective and design conditions.

Control objective. For the MAS (2.1)–(2.2), the control objective is to design a distributed output feedback MRAC law solely using local input and output information so that the closed-loop system is stable and of the higher-order output consensus properties:

(2.3)
$$\lim_{t \to \infty} (y_i(t) - y_0(t))^{(j)} = 0, \quad i = 1, \dots, N, \ j = 0, \dots, n^*,$$

where $y^{(j)}(t)$ denotes the *j*th derivative of y(t).

Assumptions. To meet the control objective given by (2.3), we present the following assumptions:

- (A1) All $Z_i(s)$, i = 1, ..., N, are stable polynomials.
- (A2) The relative degree of *i*th follower is $n_i m_i = n^*$ for i = 1, ..., N.
- (A3) An upper bound on n_i , denoted as \bar{n} , is known.
- (A4) The leader input r(t) satisfies $\dot{r}(t) \in L^{\infty}$.
- (A5) The directed graph \mathcal{G} has at least one spanning tree with v_0 being the parent.

JIAN GUO, YANJUN ZHANG, AND JI-FENG ZHANG

It is well known that the usual MRAC systems require the zeros of the control system to be stable, which is a consequence of zero-pole cancellations occurring in the MRAC systems. In this case, the MRAC law will cancel and replace the control system's zeros with the reference model's. For stability, such cancellations must be stable. In other words, the control system must be minimum-phase. Moreover, the control system's relative degree must equal the reference system's degree to guarantee model matching, which is necessary for tracking target even when the system parameters are known [37]. For a distributed MRAC design, assumptions (A1)–(A2) are regarded as extensions of the minimum-phase condition and the model-matching condition in the usual MRAC systems. Moreover, assumption (A3) is required for constructing a parameterized system model for parameter adaptation. Besides, assumptions (A1)-(A3) are the traditional design conditions in the usual MRAC systems, and assumption (A4) is a relaxed design condition on the reference system, which is used to ensure higher-order output consensus. Finally, assumption (A5) is a typical design condition for the output consensus control that is commonly used in the literature.

2.3. Comparisons and technical issues to be solved.

Comparison to cooperative output regulation. The linear cooperative output regulation problem was first formally formulated and solved using a distributed observer approach on a static network in [32] and then on a jointly connected switched network in [33]. In order to address the design condition where each follower possesses knowledge of the leader's system matrix, the literature [2] investigates the linear cooperative output regulation problem on static networks using an adaptive distributed observer approach. The output regulation based cooperative control has been systematically studied in the control community. Generally speaking, the standard output regulation for the regulator equations, which fundamentally distinguishes it from the well-known MRAC technique. This is the reason why the establishment of a fully distributed output feedback MRAC framework for cooperative control remains an imperative, necessitating our attention and focus.

Comparison to distributed MRAC. As mentioned in the introduction, distributed MRAC methods are now applied to multiagent linear time-invariant systems. However, the existing literatures [5, 21, 30, 47, 50] mainly focus on the MASs described by the state feedback for state tracking. The followers' models are of the basic form: $\dot{x}_i = A_i x_i + B_i u_i$, i = 1, ..., N, where $x_i \in \mathbb{R}^{n_i}$ and $u_i \in \mathbb{R}^{m_i}$, i = 1, ..., N, are the state vectors and input vectors of the followers, A_i and B_i , i = 1, ..., N, are unknown constant matrices of appropriate dimensions. The leader model is of the basic form: $\dot{x}_0 = A_0 x_0 + B_0 u_0$, where $x_0 \in \mathbb{R}^n$ is the state vector, $u_0 \in \mathbb{R}^m$ is the bounded reference input, and A_0 and B_0 are constant matrices, with A_0 being stable.

The control objective is to find a distributed MRAC law that ensures closed-loop stability and asymptotic state consensus $\lim_{t\to\infty} (x_i(t) - x_0(t)) = 0$. To achieve the control objective, an essential condition, known as the structural matching condition, is as follows. (i) For each follower v_i , there exists a constant matrix K_{1ij}^* and a nonsingular constant matrix K_{4i}^* of appropriate dimensions such that

(2.4)
$$A_{ei} = A_i + B_i K_{1ij}^{*T}, \quad B_{ei} = B_i K_{4i}^{*},$$

where A_{ei} is a stable and known matrix, and B_{ei} is a known matrix for i = 1, ..., N. (ii) For each pair of $(v_i, v_j) \in \mathcal{E}$, there exist constant matrices K_{2ij}^* and K_{3ij}^* of appropriate dimensions such that for i = 1, ..., N,

Copyright (c) by SIAM. Unauthorized reproduction of this article is prohibited.

(2.5)
$$A_j = A_i + B_i K_{3ij}^{*T}, \quad B_j = B_i K_{2ij}^{*}$$

The readers can refer to [30] for further details on the matching condition (2.4)–(2.5). Note that state consensus is a strong control objective. When state consensus is achieved, the followers can track the arbitrary behaviors of the leader, which requires structural similarities among all agents. Such structural similarities are modeled as the matching condition (2.4)–(2.5). However, the latter condition is restrictive for many applications, and largely restricts the application range of the consensus methods.

Technical issues to be solved. Considering that it is sufficient to achieve output consensus for more applications, this paper focuses on addressing how to develop a fully distributed output feedback MRAC scheme to ensure asymptotic output consensus for the MAS (2.1)–(2.2) without requiring the restrictive matching conditions just like (2.4)–(2.5). The basic idea of MRAC is to design an adaptive control law that ensures the closed-loop system matches any given reference system. Inspired by this, for the distributed output feedback MRAC, the agents that are connected to the leader follow the reference system (i.e., the leader model). However, the agents that are not connected to the leader do not have an available reference system. Thus, the first technical problem is designing virtual reference systems for the agents, especially for those not connected to the leader. Then, a potentially arising question is how to guarantee that the agents with virtual reference systems can achieve leader-follower output consensus. Moreover, the third technical problem is accomplishing the higherorder tracking properties (2.3). In a word, to establish a fully distributed output feedback MRAC framework, the following technical problems must be solved:

- (i) How to design the virtual reference models for all followers and construct the plant-model matching equations, especially those that are not connected to the leader, by solely using the local input and output information?
- (ii) Given that the agents could follow the virtual reference systems asymptotically, how to eventually realize leader-follower output consensus for the whole MAS (2.1)–(2.2)? Especially, asymptotic output consensus is required, which leads to more difficulties for adaptive control design and analysis.
- (iii) The current results of the distributed leader-follower control indicate that the asymptotic state/output consensus property can be ensured. However, under the usual design conditions, how to ensure some higher-order output consensus as shown in (2.3)? To the best of our knowledge, this problem has never been addressed in the literature.

3. Distributed output feedback MRC design. This section provides the basic idea of the distributed output feedback MRAC framework through a distributed model reference control (MRC) design, assuming all system parameters are known. The design contains four steps: (i) deriving the distributed MRC law structure, (ii) constructing virtual reference inputs, (iii) calculating the control law parameters, and (iv) conducting system performance analysis.

Step 1: Distributed MRC law structure. Given that all system parameters are known, we design the distributed MRC law for the *i*th agent, i = 1, ..., N, as

(3.1)
$$u_i(t) = \theta_{1i}^{*T} \omega_{1i}(t) + \theta_{2i}^{*T} \omega_{2i}(t) + \theta_{3i}^* \omega_{3i}(t) + \theta_{20i}^* y_i(t),$$

where $\theta_{1i}^* \in \mathbb{R}^{\bar{n}-1}, \theta_{2i}^* \in \mathbb{R}^{\bar{n}-1}, \theta_{3i}^* \in \mathbb{R}$, and $\theta_{20i}^* \in \mathbb{R}$ are constant parameters to be specified, and

(3.2)
$$\omega_{1i}(t) = \frac{a(s)}{\Lambda_{ci}(s)} [u_i](t) \in \mathbb{R}^{\bar{n}-1}, \quad \omega_{2i}(t) = \frac{a(s)}{\Lambda_{ci}(s)} [y_i](t) \in \mathbb{R}^{\bar{n}-1},$$

with $a(s) = [1, s, \ldots, s^{\bar{n}-2}]^T \in \mathbb{R}^{\bar{n}-1}$ and $\Lambda_{ci}(s) = s^{\bar{n}-1} + \lambda_{i,\bar{n}-2}^c s^{\bar{n}-2} + \cdots + \lambda_{i1}^c s + \lambda_{i0}^c$ representing an arbitrary monic Hurwitz polynomial. The signals $\omega_{1i}(t)$ and $\omega_{2i}(t)$ are obtained through filtering $u_i(t)$ and $y_i(t)$ by the stable filter $\frac{a(s)}{\Lambda_{ci}(s)}$, respectively.

Remark 3.1. Since $\Lambda_{ci}(s)$ in (3.2) is monic and of degree $\bar{n} - 1$ and the maximum degree of the vector a(s) is $\bar{n} - 2$, each element of the vector $\frac{a(s)}{\Lambda_{ci}(s)}$ is strictly proper, i.e., the degree of the numerator a(s) is strictly less than that of the denominator $\Lambda_{ci}(s)$. Thus, there does not exist any algebraic loop in the control law (3.1).

In traditional MRAC, $\omega_{3i}(t)$ corresponds to the reference system input. Since each agent receives signals from its neighbors, and the number of neighbors N_i is known, we design $\omega_{3i}(t)$ as

(3.3)
$$\omega_{3i}(t) = \begin{cases} \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} r_j(t), & v_0 \notin \mathcal{N}_i, \\ r(t), & v_0 \in \mathcal{N}_i, \end{cases}$$

where $r_j(t)$, $j = 1, ..., N_i$, are auxiliary signals to be designed.

From (3.3), for agents connected to the leader, the leader's input r(t) is directly used as $\omega_{3i}(t)$, enabling them to follow the leader as in traditional MRAC. For agents not connected to the leader, r(t) is unavailable. To solve this, we design the auxiliary signal $\frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} r_j(t)$ as $\omega_{3i}(t)$, which acts as a virtual reference. Designing this virtual reference and ensuring all agents can follow the leader are key challenges addressed in this paper. Next, we explain how to obtain $r_j(t)$ to construct $\omega_{3i}(t)$.

Step 2: Virtual reference input construction. As mentioned in Appendix A, traditional model reference control requires an additional reference signal $r(t) = P_m(s)[y_m](t)$, which is the sum of some derivative information of the tracked signal. Inspired by this, if the derivatives $y_j^{(k)}(t)$, $k=1,\ldots,n^*$, with respect to the *j*th agent are known, we design $r_j(t)$ as

(3.4)
$$r_i(t) = \Psi(s)[y_i](t)$$

with $\Psi(s) = s^{n^*} + \psi_{n^*-1}s^{n^*-1} + \cdots + \psi_1s + \psi_0$ being some chosen monic Hurwitz polynomials of degree n^* . However, $y_j^{(k)}(t)$ is generally difficult to be obtained. Hence, using (3.4) to obtain $r_j(t)$ is inappropriate. Thus, we present a construction method to obtain $r_j(t)$ using only u_j and y_j . For simplicity, we change the subscript from jto i, and define two vectors:

(3.5)
$$\theta_{pi}^* = [k_{pi}z_{i0}, k_{pi}z_{i1}, \dots, k_{pi}z_{i,m_i-1}, k_{pi}, -p_{i0}, -p_{i1}, \dots, -p_{i,n_i-2}, -p_{i,n_i-1}]^T \in \mathbb{R}^{n_i + m_i + 1},$$

(3.6)

$$\begin{split} \phi_i(t) &= \left[\frac{1}{\Lambda_{ei}(s)}[u_i](t), \frac{s}{\Lambda_{ei}(s)}[u_i](t), \dots, \frac{s^{m_i-1}}{\Lambda_{ei}(s)}[u_i](t), \\ &\frac{s^{m_i}}{\Lambda_{ei}(s)}[u_i](t), \frac{1}{\Lambda_{ei}(s)}[y_i](t), \frac{s}{\Lambda_{ei}(s)}[y_i](t), \\ &\dots, \frac{s^{n_i-2}}{\Lambda_{ei}(s)}[y_i](t), \frac{s^{n_i-1}}{\Lambda_{ei}(s)}[y_i](t)\right]^T \in \mathbb{R}^{n_i+m_i-1}, \end{split}$$

where $\Lambda_{ei}(s) = s^{n_i} + \lambda_{i,n_i-1}^e s^{n_i-1} + \dots + \lambda_{i1}^e s + \lambda_{i0}^e$ representing an arbitrary monic Hurwitz polynomial. Then, ignoring the exponentially decaying signal, the system (2.1) can be expressed as

(3.7)
$$y_i(t) - \frac{\Lambda_{i,n_i-1}(s)}{\Lambda_{ei}(s)} [y_i](t) = \theta_{pi}^{*T} \phi_i(t)$$

with $\Lambda_{i,n_i-1}(s) = \lambda_{i,n_i-1}^e s^{n_i-1} + \dots + \lambda_{i1}^e s + \lambda_{i0}^e$. To design $r_j(t)$, we first give the following lemma demonstrating a key property of $y_i^{(j)}(t), i = 1, \dots, N, j = 1, \dots, n^*$.

LEMMA 3.2. For $y_i^{(j)}(t)$, $j = 1, ..., n^*$, it can be expressed by $y_i^{(k)}(t)$, k = 0, ..., j-1, $\frac{s^k}{\Lambda_{ei}(s)}[u_i](t)$, $k = 1 + m_i, ..., j + m_i, \theta_{pi}^*, \phi_i(t)$, and $y_i(t)$.

Proof. The proof is given in Appendix B.

Based on Lemma 3.2, we recursively obtain that $y_i^{(j)}(t)$, $j = 1, ..., n^*$, can be expressed by $\frac{s^k}{\Lambda_{ei}(s)}[u_i](t)$ for $k = 1 + m_i, ..., j + m_i$, $\theta_{pi}^*, \phi_i(t)$, and $y_i(t)$. Thus, we express $y_i^{(j)}(t), j = 1, 2, ..., n^*$, as

(3.8)
$$y_i^{(j)} = H_{ij}\left(y_i, \frac{s^{1+m_i}}{\Lambda_{ei}(s)}[u_i], \dots, \frac{s^{j+m_i}}{\Lambda_{ei}(s)}[u_i], \theta_{pi}^*, \phi_i\right)$$

As demonstrated in the proof of Lemma 3.2, H_{ij} is obtained by applying a filter related to $\Lambda_{ei}(s)$ to the original input-output system. Its form depends solely on $\Lambda_{ei}(s)$. If $\Lambda_{ei}(s)$ is predetermined, then H_{ij} is a known mapping. Consequently, $H_{ij}, i = 1, \ldots, N, j = 1, \ldots, n^*$, are known and smooth mappings with respect to its variables. It should be noted that from (3.4), we derive an analytical expression for $r_i(t)$ as

(3.9)
$$r_{i} = \sum_{j=0}^{n^{*}} \psi_{j} H_{ij} \left(y_{i}, \frac{s^{1+m_{i}}}{\Lambda_{ei}(s)} [u_{i}], \dots, \frac{s^{j+m_{i}}}{\Lambda_{ei}(s)} [u_{i}], \theta_{pi}^{*}, \phi_{i} \right),$$

where ψ_k , $k = 1, ..., n^*$, are constant parameters with $\psi_{n^*} = 1$ such that $s^{n^*} + \psi_{n^*-1}s^{n^*-1} + \cdots + \psi_1s + \psi_0$ is a Hurwitz polynomial.

Remark 3.3. From (3.9), we see that $r_i(t)$ depends on the unknown vector $\theta_{p_i}^*$. For the adaptive control case, we construct an estimate of $r_i(t)$ that will no longer depend on any unknown information (see section 4). Besides, to estimate the higherorder derivatives of $y_i(t)$, one may employ a standard high-gain differential observer [12]. Even though the high-gain observer design is simple and easy to implement, using this observer is difficult to realize asymptotic output consensus, and involves the high-gain issue. We propose a linear parametrization-based estimation method based on this consideration to derive the $r_i(t)$'s estimate and achieve the asymptotic output consensus. Finally, it is worth noting that by (3.1), (3.3), and (3.9), it is known that each agent's controller makes use of only its own and its neighbors' information and does not need the global information of the leader.

From (3.1), it is evident that the nominal control law for each follower solely relies on local input and output information, and does not depend on global leader information.

Step 3: Calculation of θ_{1i}^* , θ_{2i}^* , θ_{3i}^* , and θ_{20i}^* . Now, we construct some plantmodel output matching equations from which θ_{1i}^* , θ_{2i}^* , θ_{3i}^* , and θ_{20i}^* can be calculated.

Motivated by the usual output feedback MRC in [37], we derive the distributed version of the plant-model output matching equations as follows.

LEMMA 3.4. For the *i*th agent connected to the leader, there exist constants $\theta_{1i}^*, \theta_{2i}^*, \theta_{20i}^*, \theta_{3i}^*$ such that

(3.10)
$$\theta_{1i}^{*T} a(s) P_i(s) + \left(\theta_{2i}^{*T} a(s) + \theta_{20i}^* \Lambda_{ci}(s) \right) k_{pi} Z_i(s) = \Lambda_{ci}(s) \left(P_i(s) - k_{pi} \theta_{3i}^* Z_i(s) P_m(s) \right);$$

and for the *i*th agent not connected to the leader, there exist constants $\theta_{1i}^*, \theta_{2i}^*, \theta_{20i}^*, \theta_{3i}^*$ such that

(3.11)
$$\theta_{1i}^{*T}a(s)P_i(s) + \left(\theta_{2i}^{*T}a(s) + \theta_{20i}^{*}\Lambda_{ci}(s)\right)k_{pi}Z_i(s) \\ = \Lambda_{ci}(s)\left(P_i(s) - k_{pi}\theta_{3i}^*Z_i(s)\Psi(s)\right),$$

where a(s) and $\Psi(s)$ are defined below (3.2) and (3.4), respectively.

Proof. The proof is similar to that of Lemma A.2 in Appendix A, and thus, omitted here. For details, one may refer to [37].

Remark 3.5. These matching equations always have nontrivial analytical solutions, and one can choose the solution $\{\theta_{1i}^*, \theta_{2i}^*, \theta_{20i}^*, \theta_{3i}^*\}$ to (3.10)–(3.11) from

(3.12)
$$\theta_{1i}^{*T}a(s) = \Lambda_{ci}(s) - Q(s)Z_i(s), \ \theta_{2i}^{*T}a(s) + \theta_{20i}^*\Lambda_{ci}(s) = -\theta_{3i}^*R_i(s),$$

and $\theta_{3i}^* = \frac{1}{k_{pi}}$, where Q(s) is the quotient of $\frac{\Lambda_{ci}(s)P_m(s)}{P_i(s)}$ and $R_i(s) = \Lambda_{ci}(s)P_m(s) - Q(s)P_i(s)$ for (3.10), and Q(s) is the quotient of $\frac{\Lambda_{ci}(s)\Psi(s)}{P_i(s)}$ and $R_i(s) = \Lambda_{ci}(s)\Psi(s) - Q(s)P_i(s)$ for (3.11).

The parameters $\theta_{1i}^*, \theta_{2i}^*, \theta_{20i}^*, \theta_{3i}^*$ in Lemma 3.4 can be called distributed matching parameters, as with these parameters, the distributed MRC law (3.1) matches all followers to the leader, as shown subsequently.

Step 4: System performance analysis. To proceed, we first define the local output tracking error as

(3.13)
$$e_i(t) = y_i(t) - \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} y_j(t), \quad i = 1, \dots, N,$$

where N_i is the number of the neighbors of agent v_i . Such a local output tracking error measures the disagreement between the follower i and the average of its neighbors on the output because it is essential to characterize the consensus level of the follower and the leader. The motivation of defining such a local state tracking error is shown as follows.

LEMMA 3.6. Under assumption (A5), if $e_i(t)$ is bounded, then $y_i(t)$ is bounded for all i = 1, ..., N. Further, if for any $j = 1, ..., n^*$, $\lim_{t\to\infty} e_i^{(j)}(t) = 0$ holds (or exponentially) for all i = 1, ..., N, then $\lim_{t\to\infty} (y_i(t) - y_0(t))^{(j)} = 0$ holds (or exponentially) for all i = 1, ..., N.

Proof. Performing a proof similar to that for Lemma 4.1 in [29], one can verify this lemma. \Box

From Lemma 3.6, global higher-order leader-follower consensus properties can be achieved as long as the higher-order derivatives of all local tracking errors (3.13) converge to zero as time tends to infinity. According to this lemma, the following theorem clarifies the closed-loop stability and output consensus performance. THEOREM 3.7. Under Assumptions (A1), (A2), and (A5), the distributed MRC law (3.1) configured with $\theta_{1i}^*, \theta_{2i}^*, \theta_{20i}^*, \theta_{3i}^*$ in Lemma 3.4 ensures that all closed-loop signals are bounded and the tracking errors $y_i(t) - y_0(t)$, i = 1, ..., N, and their derivatives up to the n^{*}th order converge to zero exponentially as $t \to \infty$.

Proof. For all agents $v_i \in \{v_i : v_0 \in \mathcal{N}_i\}$, the leader v_0 can be regarded as the reference output. Thus, based on Theorem A.3 in Appendix A, one can verify that the input $u_i(t) = \theta_{1i}^{*T} \omega_{1i}(t) + \theta_{2i}^{*T} \omega_{2i}(t) + \theta_{20i}^{*} y_i(t) + \theta_{3i}^{*} r(t)$ ensures that the signals of the agent v_i are bounded, and $y_i(t) - y_0(t)$, $i = 1, \ldots, N$, and their derivatives up to the n^* th order converge to zero exponentially.

For the agent $v_i \notin \{v_i : v_0 \in \mathcal{N}_i\}$, by Lemma 3.4, we first prove that $e_i(t)$ converges to zero exponentially. Operating both sides of (3.11) on $y_i(t)$, we have

(3.14)
$$\theta_{1i}^T a(s) P_i(s)[y_i](t) + \left(\theta_{2i}^T a(s) + \theta_{20i} \Lambda_{ci}(s)\right) k_{pi} \cdot Z_i(s)[y_i](t) = \Lambda_{ci}(s) \left(P_i(s) - k_{pi} \theta_{3i} Z_i(s) \Psi(s)\right) [y_i](t).$$

Moreover, with some manipulations on (3.1), we have

$$\Lambda_{ci}(s)[u_{i}](t) = \theta_{1i}^{T}a(s)[u_{i}](t) + \theta_{2i}^{T}a(s)[y_{i}](t) + \theta_{3i}\Lambda_{ci}(s)\Psi(s) \left[\frac{1}{N_{i}}\sum_{v_{j}\in\mathcal{N}_{i}}y_{j}\right](t) (3.15) + \Lambda_{ci}(s)\theta_{20i}[y_{i}](t) + \Lambda_{ci}(s)\left[\epsilon_{\Lambda_{ci}}\right](t),$$

where $\epsilon_{\Lambda_{ci}}(t)$ is an exponentially decaying signal associated with the initial conditions. Then, we have

$$k_{pi}Z_{i}(s)\Lambda_{ci}(s) [u_{i}](t) = P_{i}(s)\Lambda_{ci}(s) [y_{i}](t)$$

$$= k_{pi}Z_{i}(s)\Lambda_{ci}(s)\theta_{20i} [y_{i}](t) + k_{pi}Z_{i}(s)\Lambda_{ci}(s) [\epsilon_{\Lambda_{ci}}](t)$$

$$+ k_{pi}Z_{i}(s)\theta_{3i}\Lambda_{ci}(s)\Psi_{i}(s) \left[\frac{1}{N_{i}}\sum_{v_{j}\in\mathcal{N}_{i}}y_{j}\right](t)$$

$$+ k_{pi}Z_{i}(s) \left(\theta_{1i}^{T}a(s) [u_{i}](t) + \theta_{2i}^{T}a(s) [y_{i}](t)\right).$$
(3.16)

Combining (3.16) and (3.14), together with $P_i(s)[y_i](t) = k_{pi}Z_i(s)[u_i](t)$, indicates that

(3.17)
$$\Lambda_{ci}(s)\Psi(s)Z_i(s)[y_i - \frac{1}{N_i}\sum_{v_j \in \mathcal{N}_i} y_j](t) = -k_{pi}Z_i(s)\Lambda_{ci}(s)\left[\epsilon_{\Lambda_{ci}}\right](t).$$

Since $\Lambda_{ci}(s)$, $\Psi(s)$, and $Z_i(s)$ are all stable polynomials and the degree of $\Psi(s)$ is n^* , we conclude that for $l = 0, 1, ..., n^*$,

(3.18)
$$\left(y_i(t) - \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} y_j(t)\right)^{(l)} \to 0, \text{ exponentially.}$$

According to Lemma 3.6, (3.18) suggests that the higher order exponential leaderfollower consensus (2.3) is achieved. This also implies that $y_i(t) \in L^{\infty}$ due to the boundedness of $y_0(t)$.

Now, we prove $u_i(t)$, i = 1, ..., N, are also bounded. Using (2.1) and (3.17), we have $k_{pi}Z_i(s)^2 \Lambda_{ci}(s)\Psi(s)[u_i(t)] = P_i(s)\Lambda_{ci}(s)Z_i(s)[\frac{1}{n_i}\sum_{v_j\in\mathcal{N}_i}r_j](t) + \epsilon_{1i}(t)$ with

 $\epsilon_{1i}(t) = -k_{pi}Z_i(s)\Lambda_{ci}(s) [\epsilon_{\Lambda_{ci}}](t)$. Since $\Lambda_{ci}(s), \Psi(s)$, and $Z_i(s)$ are all stable, we can derive

$$u_i(t) = \frac{P_i(s)}{k_{pi}Z_i(s)\Psi_i(s)} \left[\frac{1}{N_i}\sum_{v_j \in \mathcal{N}_i} r_j\right](t) + \epsilon_{2i}(t),$$

where $\epsilon_{2i}(t)$ is an exponentially decaying signal associated with initial conditions. Note that $\frac{P_i(s)}{k_{pi}Z_i(s)\Psi(s)}$ is stable and proper, i.e., the degree of the numerator $P_i(s)$ is not greater than that of the denominator $k_{pi}Z_i(s)\Psi(s)$. Thus, if $\sum_{v_i\in\mathcal{N}_i}r_j\in L^{\infty}$, then $u_i(t) \in L^\infty$.

Let l_i denote the length of the longest directed path for the leader v_0 to the node v_i . Suppose that there exists a follower v_k such that r_k is unbounded. Then, there exists a neighbor v_{k_i} of v_k such that r_{k_i} is unbounded and $l_{k_i} < l_k$. From assumption (A5), and by repeating this analysis for up to l_k steps, we conclude that the reference signal of the leader r(t) is unbounded, which is a contradiction. Therefore, $r_i(t) \in L^{\infty}$, $i = 1, \ldots, N$, and so are the control $u_i(t)$. This completes the proof. Π

Remark 3.8. Equation (3.17) shows that the convergence rate is influenced by the roots of a certain polynomial, with larger roots leading to faster convergence speed. However, large roots can cause initial output fluctuations. Therefore, the choice of Λ_{ei} and Λ_{ci} should consider both the convergence speed and the transient performance of the system.

So far, we have provided a basic distributed MRC framework for the MAS (2.1)-(2.2) which is fundamental for the *distributed MRAC design* addressed next.

4. Distributed output feedback MRAC design. This section develops a distributed output feedback indirect MRAC scheme for the MAS (2.1)–(2.2), where the parameters p_{ij}, z_{ij} , and k_{pi} are unknown. Specifically, we construct the distributed output feedback MRAC law, with the distributed indirect MRAC design procedure comprising five steps: (i) distributed MARC law construction, (ii) plant parameter estimation, (iii) controller parameter calculation, (iv) virtual reference input signal estimation, and (v) stability performance analysis.

Step 1: Distributed MARC law structure. The distributed MRAC law is designed as

(4.1)
$$u_i(t) = \theta_{1i}^T(t)\omega_{1i}(t) + \theta_{2i}^T(t)\omega_{2i}(t) + \theta_{3i}(t)\hat{\omega}_{3i}(t) + \theta_{20i}(t)y_i(t),$$

where $\theta_{1i}(t)$ and $\theta_{2i}(t)$ are estimates of θ_{1i}^* and θ_{2i}^* in Lemma 3.4, respectively, $\theta_{3i}(t)$ is an estimate of $\frac{1}{k_{pi}}$, $\omega_{1i}(t)$ and $\omega_{2i}(t)$ are defined in (3.2), and $\hat{\omega}_{3i}(t)$ is an estimate of $\omega_{3i}(t)$ in (3.3).

Step 2: Plant parameter estimation. Consider the *i*th follower in (2.1). The signal $\phi_i(t)$ in (3.6) can be obtained through filtering $u_i(t)$ and $y_i(t)$ by the stable filter $\frac{a_i(s)}{\Lambda_{ei}(s)}$ with $a_i(s) = \begin{bmatrix} 1, s, \dots, s^{n_i-2} \end{bmatrix}^T$ and $\Lambda_{ei}(s)$ below (3.6). Similarly, $\frac{\Lambda_{i,n_i-1}(s)}{\Lambda_{ei}(s)} [y_i](t)$ $\Lambda_{ei}(s)$ (1.1) $\Lambda_{i}(s)$ $\Gamma_{i}(s)$ $\Gamma_{i}(s)$ in (3.7) can be obtained through filtering $y_i(t)$ by the stable filter $\frac{\Lambda_{i,n_i-1}(s)}{\Lambda_{ei}(s)}$

Let $\theta_{pi}(t)$ be an estimate of θ_{pi}^* and define the estimation error as

(4.2)
$$\epsilon_i(t) = \theta_{pi}^T(t)\phi_i(t) - y_i(t) + \frac{\Lambda_{i,n_i-1}(s)}{\Lambda_{ei}(s)}[y_i](t), \quad t \ge t_0.$$

 \mathbf{n} () ()

To update $\theta_{pi}(t)$, we use the following gradient algorithm:

(4.3)
$$\dot{\theta}_{pi}(t) = -\frac{\Gamma_i \phi_i(t) \epsilon_i(t)}{m_i^2(t)}, \quad \theta_{pi}(t_0) = \theta_{0i}, \quad t \ge t_0,$$

where $\Gamma_i = \text{diag} \{\Gamma_{1i}, \gamma_{m_i+1}, \Gamma_{2i}\}$ with $\Gamma_{1i} \in \mathbb{R}^{m_i \times m_i}, \Gamma_{1i} = \Gamma_{1i}^T > 0, \gamma_{m_i+1} > 0$, and $\Gamma_{2i} \in \mathbb{R}^{n_i \times n_i}, \Gamma_{2i} = \Gamma_{2i}^T > 0, \theta_{0i}$ is an initial estimate of $\theta_{pi}^* \in \mathbb{R}^{n_i+m_i+1}$, and

(4.4)
$$m_i(t) = \sqrt{1 + \kappa \phi_i^T(t)\phi_i(t)}, \ \kappa > 0.$$

From (3.5), we denote $\theta_{pi}(t)$ as

$$\theta_{pi}(t) = \left[\widehat{k_{pi}z_{i0}}(t), \dots, \widehat{k_{pi}z_{im,i-1}}(t), \hat{k}_{pi}(t), -\hat{p}_{i0}(t), \dots, -\hat{p}_{i,n_i-1}(t)\right]^T$$

Thus, we construct the estimates of $P_i(s)$ and $Z_i(s)$ for the *i*th follower as

(4.5)
$$P_i(s, \hat{p}_i) = s^{n_i} + \hat{p}_{i,n_i-1}s^{n_i-1} + \dots + \hat{p}_{i1}s + \hat{p}_{i0},$$
$$\hat{Z}_i(s, \hat{z}_i) = s^{m_i} + \hat{z}_{i,m_i-1}s^{m_i-1} + \dots + \hat{z}_{i1}s + \hat{z}_{i0},$$

where $\hat{z}_i = [\hat{z}_{i0}, \ldots, \hat{z}_{i,m_i-1}]^T$ with $\hat{z}_{ij} = \frac{\widehat{k_{pi}z_{ij}}(t)}{\widehat{k}_{pi}(t)}$ and $\hat{p}_i = [\hat{p}_{i0}, \ldots, \hat{p}_{i,n_i-1}]^T$ are the estimates of $z_i^* = [z_{i0}, \ldots, z_{i,m_i-1}]^T$ and $p_i^* = [p_{i0}, \ldots, p_{i,n_i-1}]^T$, respectively. To ensure $\hat{k}_{pi}(t) \neq 0$ during parameter adaptation, the parameter update law (4.3) needs to be modified by introducing some robust term, such as parameter projection, dead-zone modification, σ -modification, and so on. We omit the details due to the paper length constraints.

For the parameter $\theta_{pi}(t)$, the following lemma clarifies some properties crucial for stability analysis.

LEMMA 4.1. The adaptive algorithm (4.3) guarantees (i) $\theta_{pi}(t), \dot{\theta}_{pi}(t), \frac{\epsilon_i(t)}{m_i(t)}$ are bounded and (ii) $\frac{\epsilon_i(t)}{m_i(t)}$ and $\dot{\theta}_{pi}(t)$ belong to L^2 .

Proof. The proof is similar to Lemma 3.1 in [37], and so, it is omitted here.

Note that the regressor vector $\phi_i(t)$ is not required to be persistently exciting, and thus, we cannot ensure that the estimation errors $\epsilon_i(t)$ converge to zero. Nevertheless, this paper shows that the proposed distributed MRAC law (4.1) still ensures closed-loop stability and the tracking properties shown in (2.3).

Step 3: Controller parameter calculation. For the *i*th agent connected to the leader, the controller parameters $\{\theta_{1i}(t), \theta_{2i}(t), \theta_{20i}(t), \theta_{3i}(t)\}$ are obtained from

(4.6)
$$\theta_{1i}^{T}a(s)\hat{P}_{i}(s,\hat{p}_{i}) + \left(\theta_{2i}^{T}a(s) + \theta_{20i}\Lambda_{ci}(s)\right)k_{pi}\hat{Z}_{i}(s,\hat{z}_{i}) = \Lambda_{ci}(s)\left(\hat{P}_{i}(s,\hat{p}_{i}) - \hat{k}_{pi}\theta_{3i}\hat{Z}_{i}(s,\hat{z}_{i})P_{m}(s)\right),$$

and for the ith agent not connected to the leader, the controller parameters are obtained from

(4.7)
$$\theta_{1i}^{T}a(s)\hat{P}_{i}(s,\hat{p}_{i}) + \left(\theta_{2i}^{T}a(s) + \theta_{20i}\Lambda_{ci}(s)\right)k_{pi}\hat{Z}_{i}(s,\hat{z}_{i})$$
$$= \Lambda_{ci}(s)\left(\hat{P}_{i}(s,\hat{p}_{i}) - \hat{k}_{pi}\theta_{3i}\hat{Z}_{i}(s,\hat{z}_{i})\Psi(s)\right).$$

Regarding how to specifically derive $\theta_{1i}(t)$, $\theta_{2i}(t)$, $\theta_{20i}(t)$, $\theta_{3i}(t)$, the reader can refer to (3.12).

Step 4: Virtual reference input signal estimation. The signal $\hat{\omega}_{3i}(t)$ in (4.1) is designed by

(4.8)
$$\hat{\omega}_{3i}(t) = \begin{cases} \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} \hat{r}_j(t), & v_0 \notin \mathcal{N}_i, \\ r(t), & v_0 \in \mathcal{N}_i, \end{cases}$$

where $\hat{r}_j(t)$ is an estimate of the signal $r_j(t)$. For simplicity, we change the subscript of $\hat{r}_j(t)$ from j to i, and design $\hat{r}_i(t)$ as

(4.9)
$$\hat{r}_{i} = \sum_{j=0}^{n^{*}} \psi_{j} H_{ij} \left(y_{i}, \frac{s^{1+m_{i}}}{\Lambda_{ei}(s)} [u_{i}], \dots, \frac{s^{j+m_{i}}}{\Lambda_{ei}(s)} [u_{i}], \theta_{pi}, \phi_{i} \right).$$

Now, we derive the following lemma to demonstrate a convergent property of the error $\hat{r}_i(t) - r_i(t)$ under some particular conditions.

LEMMA 4.2. For the gradient algorithm (4.3), if $m_i(t) \in L^{\infty}$, $\dot{u}_i(t) \in L^{\infty}$, and $\dot{y}_i(t) \in L^{\infty}$, then we have $\dot{r}_i(t) \in L^{\infty}$ and

(4.10)
$$\lim_{t \to \infty} (\hat{r}_i(t) - r_i(t)) = 0.$$

Proof. The proof of this lemma is long. Thus, we present it in Appendix B to avoid disrupting the reading flow. \Box

Step 5: System performance analysis. Based on the above derivations, we provide the main result of this paper, which demonstrates that the closed-loop stability and asymptotic higher-order output consensus are achieved by using the distributed MRAC law (4.1).

THEOREM 4.3. Under assumptions (A1)–(A5), the distributed output feedback MRAC law (4.1) ensures that all signals in the adaptive control system comprising (2.1), (2.2), (4.1), and (4.3) are bounded, and for i = 1, ..., N,

(4.11)
$$\lim_{t \to \infty} (y_i(t) - y_0(t))^{(k)} = 0, \quad k = 0, \dots, n^*$$

Proof. First, we prove that the agents connected to the leader can track the leader and generate a virtual signal $\hat{r}(t)$ satisfying $\lim_{t\to\infty}(\hat{r}(t)-r(t)) \to 0$ and $\dot{r}(t) \in L^{\infty}$. For the *i*th agent connected to the leader, the control law becomes $u_i(t) = \theta_{1i}^T(t)\omega_{1i}(t) + \theta_{2i}^T(t)\omega_{2i}(t) + \theta_{3i}(t)r(t) + \theta_{20i}(t)y_i(t)$. Hence, from Theorem A.4 in Appendix A, we have the closed-loop stability and $\lim_{t\to\infty}(y_i(t)-y_0(t)) = 0$. Under assumption (A4), we have $\dot{u}_i(t) \in L^{\infty}$ and $\dot{y}_i(t) \in L^{\infty}$. Following Lemma 4.2, and combined with the closed loop stability yields $\lim_{t\to\infty}(\hat{r}_i(t)-r(t)) = 0$ and $\dot{r}_i(t) \in L^{\infty}$.

Second, we prove that for the *i*th agent, if the conditions $\lim_{t\to\infty} (\hat{r}_j(t) - r_j(t)) = 0$ and $\dot{r}_j(t) \in L^{\infty}$ are satisfied for any $v_j \in \mathcal{N}_i$, then the following properties hold:

(4.12)
$$\lim_{t \to \infty} \left(y_i(t) - \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} y_j(t) \right)^{(k)} = 0,$$

for any $k = 0, ..., n^*, i = 1, ..., N$, and $\dot{r}_i(t) \in L^{\infty}$. In view of the control (4.1), for any $v_j \in \mathcal{N}_i$, define

(4.13)
$$\hat{y}_j(t) = \frac{1}{\Psi(s)} [\hat{r}_j](t).$$

Then, ignoring the exponentially decaying signal, it follows from (4.13) that $\hat{r}_j(t) = \Psi(s)[\hat{y}_j](t)$. Substituting it into (4.8) yields $\hat{\omega}_{3i}(t) = \Psi(s)[\frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} \hat{y}_j](t)$. Based on Theorem A.4 in Appendix A with $u_i(t) = \theta_{1i}^T(t)\omega_{1i}(t) + \theta_{2i}^T(t)\omega_{2i}(t) + \theta_{3i}(t)\hat{\omega}_{3i}(t) + \theta_{20i}(t)y_i(t)$, all signals with respect to the *i*th agent system are bounded and $\lim_{t\to\infty}(y_i(t) - \frac{1}{N_i}\sum_{v_j\in\mathcal{N}_i}\hat{y}_j(t)) = 0$. Moreover, we further verify that

DISTRIBUTED OUTPUT FEEDBACK INDIRECT MRAC

(4.14)
$$\lim_{t \to \infty} \left(y_i(t) - \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} \hat{y}_j \right)^{(k)} = 0, \quad k = 0, \dots, n^*.$$

Proving (4.14) is quite similar to that of Theorem 3.1 in [38], and thus, omitted here. Since

(4.15)

$$\lim_{t \to \infty} \left(y_i(t) - \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} y_j(t) \right)^{(k)}$$
$$= \lim_{t \to \infty} \left(y_i(t) - \frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} \hat{y}_j(t) \right)^{(k)} + \lim_{t \to \infty} \left(\frac{1}{N_i} \sum_{v_j \in \mathcal{N}_i} \left(\frac{1}{\Psi(s)} [\hat{r}_j - r_j](t) \right)^{(k)} \right)$$

it is sufficient to prove that for any $v_j \in \mathcal{N}_i$, the following equation holds:

(4.16)
$$\lim_{t \to \infty} \left(\frac{1}{\Psi(s)} \left[\hat{r}_j - r_j \right](t) \right)^{(k)} = 0$$

Let $\varepsilon_j(t) = \hat{r}_j(t) - r_j(t)$ and the *k*th order time derivative of $\frac{1}{\Psi(s)}[\varepsilon_j](t)$ is $\frac{s^k}{\Psi(s)}[\varepsilon_j](t)$. Thus, with $\frac{s^k}{\Psi(s)}$ being stable and proper, if $\lim_{t\to\infty}(\hat{r}_j(t) - r_j(t)) = 0$ for $v_j \in \mathcal{N}_i$, the property (4.16) holds. Moreover, if $\dot{\hat{r}}_j(t) \in L^\infty$ for $v_j \in \mathcal{N}_i$, then $\dot{u}_i(t) \in L^\infty$ and $\dot{y}_i(t) \in L^\infty$. From Lemma 4.2, it follows that $\dot{\hat{r}}_i(t) \in L^\infty$.

Third, we prove that $\lim_{t\to\infty} (\hat{r}_i(t) - r_i(t)) = 0$ and $\hat{r}_i(t) \in L^{\infty}$ for $i = 1, \ldots, N$. We demonstrate that each agent satisfies $\dot{r}_i(t) \in L^{\infty}$. Let l_i denote the length of the longest directed path for the leader v_0 to the node v_i . Suppose there exists at least one agent v_k such that $\dot{r}_k(t)$ is unbounded. Then, there exists a neighbor v_{k_j} of v_k such that \dot{r}_{k_j} is unbounded and $l_{k_j} < l_k$. Repeating this analysis for up to l_k steps, it concludes that the reference signal of the leader $\dot{r}(t)$ is unbounded, which contradicts assumption (A5). Therefore, $\hat{r}_i(t) \in L^{\infty}$, $i = 1, \ldots, N$. Then, we get $m_i(t) \in L^{\infty}$, $\dot{u}_i(t) \in L^{\infty}$, and $\dot{y}_i(t) \in L^{\infty}$ and Lemma 4.2 indicates $\lim_{t\to\infty} (\hat{r}_i(t) - r_i(t)) = 0$ and $\dot{r}_i(t) \in L^{\infty}$.

Finally, we demonstrate the tracking convergence and the higher-order properties. From the second and third steps, we get $\lim_{t\to\infty}(y_i(t) - \frac{1}{N_i}\sum_{v_j\in\mathcal{N}_i}y_j(t))^{(k)} = 0$, for any $k = 0, \ldots, n^*, i = 1, \ldots, N$. This, together with Lemma 3.6, indicates that $\lim_{t\to\infty}(y_i(t) - y_0(t))^{(k)} = 0$ for all $k = 0, \ldots, n^*$ and $i = 1, \ldots, N$. The proof is completed.

Remark 4.4. Theorem 4.3 addresses the tracking performance in the presence of unknown parameters. If the reference signal $r_0(t)$ meets certain additional conditions, such as being sufficiently rich of order $2\bar{n}$, then the tracking error can further converge to zero exponentially. For more details, please refer to reference [10].

So far, we have established a fully distributed output feedback MRAC scheme, where the adaptive control law for each follower only relies on its local input and output information, and the asymptotic leader-follower output consensus is achieved. Particularly, the proposed adaptive control scheme overcomes the restrictive structural matching conditions, e.g., (2.4) and (2.5), commonly used in the existing distributed MRAC literature. Moreover, the higher-order leader-follower output consensus is achieved without using the persistent excitation condition as shown in Theorem 4.3.



FIG. 1. Communication graph for nominal control design.

5. Simulation examples. This section presents an example to demonstrate the design procedure and verify Theorem 3.7, Lemma 4.2, and Theorem 4.3. We study the consensus performance of four followers and a virtual leader for the nominal control case and adaptive control case, and their associated communication graph is shown in Figure 1.

Simulation system. Consider the following MAS containing four followers modeled as

(5.1)
$$P_i(s)[y_i](t) = k_{pi}Z_i(s)[u_i](t), \quad t \ge 0, \quad i = 1, 2, 3, 4$$

where $P_1(s) = (s+1)(s-\frac{1}{2}), Z_1(s) = s+\frac{1}{2}, P_2(s) = (s+\frac{3}{2})(s-\frac{1}{2})(s+\frac{1}{2}), Z_2(s) = (s+\frac{1}{2})(s+1), P_3(s) = (s-1)(s+2), Z_3(s) = s+\frac{1}{3}, P_4(s) = (s-1)(s-\frac{1}{2})(s+2), Z_4(s) = (s+\frac{1}{3})(s+\frac{1}{4}), \text{ and } k_{p1} = -1/3, k_{p2} = 2, k_{p3} = -3, k_{p4} = 4.$ Note that the followers' models considered in this simulation are unstable and heterogeneous. The leader model is chosen as

(5.2)
$$y_0(t) = W_m(s) [r_0](t)$$

with $W_m(s) = 1/P_m(s) = \frac{1}{s+1}$ and $y_0(t) = 5\sin(2t)$. Thus, we calculate that $r(t) = 10\cos(2t) + 5\sin(2t)$.

Nominal control case. When the parameters are known, we utilize distributed MRC law to achieve convergence.

Distributed MRC law specification. Based on (3.1), the distributed MRC law for the MAS (5.1)–(5.2) is designed as

(5.3)
$$u_i(t) = \theta_{1i}^{*T} \omega_{1i}(t) + \theta_{2i}^{*T} \omega_{2i}(t) + \theta_{20i}^{*} y_i(t) + \theta_{3i}^{*} \omega_{3i}(t),$$

where $\omega_{ji}(t), j = 1, 2, 3$, can be derived from (3.2) and (3.3) with $\Lambda_{c1}(s) = s + 1, \Lambda_{c2}(s) = s^2 + 1.5s + 0.5, \Lambda_{c3}(s) = s + 1, \Lambda_{c4}(s) = s^2 + 1.5s + 0.5$, and $\Psi(s) = s + 1.5$. Moreover, by Lemma 3.4, the matching parameters in (5.3) are calculated as

$$\begin{aligned} \theta_{11}^* &= 0.5, \ \theta_{21}^* = 0, \ \theta_{201}^* = 4.5, \ \theta_{31}^* = -3, \ \theta_{12}^* = [-53.5, -53.5]^T, \\ \theta_{22}^* &= [-33.625, -13.75]^T, \ \theta_{202}^* = 26.25, \ \theta_{32}^* = 0.5, \\ \theta_{13}^* &= 0.6667, \ \theta_{23}^* = 0.6667, \ \theta_{203}^* = 0.5, \ \theta_{33}^* = -0.3333, \\ \theta_{14}^* &= [0.4167, 0.9167]^T, \ \theta_{24}^* = [0.3750, -0.3750]^T, \ \theta_{204}^* = -0.6250, \ \theta_{34}^* = 0.25. \end{aligned}$$

System responses. The initial outputs of the followers are chosen as $[y_1(0), y_2(0), y_3(0), y_4(0)]^T = [3.5, 6, 0, 8.3]^T$. Figure 2 shows the response of the outputs $y_i(t), i = 1, \ldots, 4$, of the followers and the trajectories of the derivatives of the leader and followers' output. Figure 2 highlights that the desired output higher order consensus performance is ensured. The simulation results verify the theoretical results.

Adaptive control case. To verify Lemma 4.2 and Theorem 4.3, consider the system (5.1)-(5.2) where the parameters are unknown.



FIG. 2. Trajectories of the five agents' outputs and derivatives.

Distributed MRAC law specification. Based on (4.1), the distributed MRAC law for the MAS (5.1)–(5.2) is designed as

(5.4)
$$u_i(t) = \theta_{1i}^T(t)\omega_{1i}(t) + \theta_{2i}^T(t)\omega_{2i}(t) + \theta_{20i}(t)y_i(t) + \theta_{3i}(t)\hat{\omega}_{3i}(t),$$

where $\omega_{ji}(t), j = 1, 2$, can be derived from (3.2) with $\Lambda_{c1}(s) = s + 4, \Lambda_{c2}(s) = s^2 + 5s + 6, \Lambda_{c3}(s) = s + 5, \Lambda_{c4}(s) = s^2 + 7s + 12$, and $\Psi(s) = s + 1.5$. Moreover, to obtain the adaptive parameters $\theta_{1i}(t), \theta_{2i}(t), \theta_{20i}(t), \theta_{3i}(t)$ in (5.4), first by (4.3), we obtain the estimates of θ_{pi}^* defined in (3.5) with $\Gamma_1 = \Gamma_3 = 10I_{4\times4}, \Gamma_2 = \Gamma_4 = 10I_{6\times6},$ and $\Lambda_{e1}(s) = s^2 + 3s + 2, \Lambda_{e2}(s) = s^3 + 1.833s^2 + s + 0.167, \Lambda_{e3}(s) = s^2 + 1.333s + 0.333, \Lambda_{e4}(s) = s^3 + 1.833s^2 + s + 0.167,$ where $\phi_i(t), \epsilon_i(t)$, and $m_i(t)$ can be derived from (3.6), (4.2), and (4.4), respectively. Then, $\theta_{1i}(t), \theta_{2i}(t), \theta_{20i}(t), \theta_{3i}(t)$ can be calculated by (4.6) and (4.7). Next, we specify the signal (4.8) as

$$\hat{\omega}_{31}(t) = \hat{\omega}_{32}(t) = r(t), \quad \hat{\omega}_{33}(t) = 1/2(\hat{r}_1(t) + \hat{r}_2(t)), \quad \hat{\omega}_{34}(t) = 1/2(\hat{r}_2(t) + \hat{r}_3(t)),$$

where

$$\hat{r}_j(t) = \theta_{pj}^T(t)s[\phi_j](t) + \frac{s\Lambda_{j,n-1}(s)}{\Lambda_{ej}(s)}[y_j](t) + 1.5y_j(t), \quad j = 1, 2, 3, 4,$$

with $\phi_j(t)$ defined in (3.6) and $\Lambda_{j(n-1)}(s)$ defined below (3.7).

System responses. The initial outputs of the followers are chosen as $[y_1(0), y_2(0), y_3(0), y_4(0)]^T = [-1, 2, 3, 1]^T$. Figure 3 displays the first element of the adaptive parameters $\{\theta_{1i}(t), \theta_{2i}(t), \theta_{20i}(t), \theta_{3i}(t)\}$ in (5.4) and Figure 4 presents the responses of the outputs $y_i(t), i = 1, \ldots, 4$, of the followers. Figure 5 shows the trajectories of the followers' inputs, and Figure 6 displays the consistency of the estimated virtual reference signal. From Figure 6, Lemma 4.2 is well verified. Figure 7 illustrates the trajectories of the higher-order properties in Theorem 4.3 are well supported by the numerical example. Overall, the simulation results have verified the theoretical results for the adaptive control case. Here we provide only numerical examples, while how to apply the proposed method in a real application is currently under investigation.

6. Conclusion. This paper proposes a fully distributed output feedback MRAC method for a general class of linear time-invariant systems with unknown parameters. The developed architecture overcomes the restrictive matching condition commonly used in the existing distributed MRAC methods. Our adaptive control law solely relies on local input and output information and ensures global higher-order leader-follower



FIG. 3. Trajectories of the parameter adaptation.



FIG. 4. Trajectories of the agents' outputs.



FIG. 5. Trajectories of the followers' inputs.

output consensus. Several simulation results verify the validity of the proposed adaptive control method. Nevertheless, how to solve the issues when the MAS (1)–(2) with uncertain switching topologies by using a distributed output feedback MRAC framework should be further studied.

Appendix A. Some useful lemmas and theorems. The following lemma establishes a crucial link between the square integrability property of a function and the asymptotic convergence of an associated error signal. Specifically, it states that if a function f(t) has a bounded derivative and the integral $\int_0^{\infty} f^2(t)dt$ is finite, then f(t) asymptotically approaches zero as $t \to \infty$. This lemma is a specific application of



FIG. 6. Trajectories of the followers' virtual signals.



FIG. 7. Trajectories of the agents' output derivatives.

a more general result known as Barbălat's lemma, which guarantees the convergence of certain types of functions under the given conditions [10].

LEMMA A.1. (See [37].) If $\dot{f}(t) \in L^{\infty}$ and $f(t) \in L^2$, then $\lim_{t\to\infty} f(t) = 0$.

Now we present some well-known results of traditional indirect MRAC of LTI systems, which are fundamentals in our distributed output feedback MRAC design.

Consider a traditional indirect MRAC system. The control system is

(A.1)
$$P(s)[y](t) = k_p Z(s)[u](t),$$

where y is the output, u is the input, P(s) is the pole polynomial with unknown coefficients, Z(s) is the stable zero polynomial with unknown coefficients, and k_p is the unknown high-frequency gain. The reference model is

(A.2)
$$P_m(s)[y_m](t) = r(t).$$

The indirect MRAC law is

(A.3)
$$u(t) = \theta_1^T \omega_1(t) + \theta_2^T \omega_2(t) + \theta_{20} y(t) + \theta_3 r(t),$$

where θ_i , i = 1, 2, 20, 3, are designed parameters, $\omega_1(t) = \frac{a(s)}{\Lambda_c(s)}[u](t) \in \mathbb{R}^{n-1}, \omega_2(t) = \frac{a(s)}{\Lambda_c(s)}[y](t) \in \mathbb{R}^{n-1}$ with $a(s) = [1, s, \dots, s^{n-2}]$, and $\Lambda_c(s)$ being a monic stable polynomial of degree n-1.

Copyright (c) by SIAM. Unauthorized reproduction of this article is prohibited.

LEMMA A.2. (See [37].) There exist constant parameters $\theta_1^*, \theta_2^*, \theta_{20}^*, \theta_3^*$ such that

(A.4)
$$\theta_1^{*T}a(s)P(s) + (\theta_2^{*T}a(s) + \theta_{20}^*\Lambda_c(s))Z(s) = \Lambda_c(s)(P(s) - \theta_3^*Z(s)P_m(s))$$

THEOREM A.3. (See [37].) If the parameters θ_i in (A.3) are replaced by θ_i^* , i = 1, 2, 20, 3, satisfying (A.4), then the control law (A.3) ensures that all signals in the closed-loop system are bounded and $y(t) - y_m(t) = \epsilon_0(t)$ for some initial conditionrelated exponentially decaying $\epsilon_0(t)$.

For the adaptive case, there are two steps to design θ_i , i = 1, 2, 20, 3: (i) estimation of the system parameters by an adaptive law like (4.3), and (ii) calculation of the controller parameters using some linear equations like (31). Under some standard assumptions, the indirect MRAC system (A.1)-(A.3) has the following properties. All these properties can be seen in [37].

THEOREM A.4. (See [37].) The adaptive control law (A.3) ensures that all signals are bounded and $y(t) - y_m(t) \in L^2$, $\lim_{t\to\infty} (y(t) - y_m(t)) = 0$.

Appendix B. Proofs of Lemmas 3.2 and 4.2.

B.1. Proof of Lemma 3.2. Using $\Lambda_{ei}(s)$ defined below (3.6), we can express the agent model (1) of the following form:

(B.1)
$$y_i(t) - \frac{\Lambda_{i(n_i-1)}(s)}{\Lambda_{ei}(s)} [y_i](t) = \theta_{pi}^{*T} \phi_i(t).$$

Then, we have

(B.2)
$$s[y_i](t) = \theta_{pi}^{*T} s[\phi_i](t) + \frac{s\Lambda_{i(n_i-1)}(s)}{\Lambda_{ei}(s)} [y_i](t) \\ = \theta_{pi}^{*T} \left[\frac{s}{\Lambda_{ei}(s)} [u_i](t), \dots, \frac{s^{m_i+1}}{\Lambda_{ei}(s)} [u_i](t), \\ \frac{s}{\Lambda_{ei}(s)} [y_i](t), \dots, \frac{s^{n_i}}{\Lambda_{ei}(s)} [y_i](t) \right]^T + \frac{s\Lambda_{i(n_i-1)}(s)}{\Lambda_{ei}(s)} [y_i](t).$$

Since the degree of $\Lambda_{ei}(s)$ is n_i , then $\frac{s}{\Lambda_{ei}(s)}[u_i](t), \ldots, \frac{s^{m_i+1}}{\Lambda_{ei}(s)}[u_i](t)$ and $\frac{s}{\Lambda_{ei}(s)}[y_i](t)$, $\ldots, \frac{s^{n_i-1}}{\Lambda_{ei}(s)}[y_i](t)$ can be expressed by $\phi_i(t)$. Moreover, we calculate

$$\frac{s^{n_i}}{\Lambda_{ei}(s)}[y_i](t) = y_i(t) + \frac{s^{n_i} - \Lambda_{ei}(s)}{\Lambda_{ei}(s)}[y_i](t),$$

$$\frac{s\Lambda_{i(n_i-1)}(s)}{\Lambda_{ei}(s)}[y_i](t) = \Lambda^e_{i(n_i-1)}y_i(t) + \frac{s\Lambda_{i(n_i-1)}(s) - \Lambda^e_{i(n_i-1)}\Lambda_{ei}(s)}{\Lambda_{ei}(s)}[y_i](t),$$

where $\frac{s^{n_i} - \Lambda_{ei}(s)}{\Lambda_{ei}(s)}$, and $\frac{s\Lambda_{i(n_i-1)}(s) - \Lambda_{ei}(s)}{\Lambda_{ei}(s)}$ are strictly proper. This indicates that Lemma 3.2 holds for j = 1.

When $1 < j < n^*$, we have

(

$$s^{j}[y_{i}](t) = \theta_{pi}^{*T} s^{j}[\phi_{i}](t) + \frac{s^{j} \Lambda_{i(n_{i}-1)}(s)}{\Lambda_{ei}(s)} [y_{i}](t)$$

$$= \theta_{pi}^{*T} \left[\frac{s^{j}}{\Lambda_{ei}(s)} [u_{i}](t), \dots, \frac{s^{m_{i}+j}}{\Lambda_{ei}(s)} [u_{i}](t) \frac{s^{j}}{\Lambda_{ei}(s)} [y_{i}](t), \dots, \frac{s^{n_{i}-1+j}}{\Lambda_{ei}(s)} [y_{i}](t) \right]^{T}$$

B.3)
$$+ \frac{s^{j} \Lambda_{i(n_{i}-1)}(s)}{\Lambda_{ei}(s)} [y_{i}](t).$$

Noting that $j < n^*$, $n_i = m_i + n^*$, the signals $\frac{s^j}{\Lambda_{ei}(s)}[u_i](t), \ldots, \frac{s^{m_i+j}}{\Lambda_{ei}(s)}[u_i](t)$, and $\frac{s^j}{\Lambda_{ei}(s)}[y_i](t), \ldots, \frac{s^{j+(n_i-1-j)}}{\Lambda_{ei}(s)}[y_i](t)$ can be directly obtained. Moreover, through decomposition, one can obtain

$$\frac{s^{n_i+q}}{\Lambda_{ei}(s)} = \sum_{k=0}^{q} \bar{h}_{qk} s^{q-k} + \sum_{k=1}^{n_i-1} \bar{l}_{qk} \frac{s^k}{\Lambda_{ei}(s)}, \quad q = 0, \dots, j-1,$$

B.4)
$$\frac{s^j \Lambda_{i(n_i-1)}(s)}{\Lambda_{ei}(s)} = \sum_{k=0}^{j-1} \check{h}_k s^{j-1-k} + \sum_{k=1}^{n_i-1} \check{l}_k \frac{s^k}{\Lambda_{ei}(s)}.$$

Thereby, $s^{j}[y_{i}](t)$, $j = 1, 2, ..., n^{*} - 1$, can be expressed by $s[y_{i}](t), ..., s^{j-1}[y_{i}](t), \theta_{pi}^{*}$ in (3.5), $\frac{s^{k}}{\Lambda_{ei}(s)}[u_{i}](t), k = 1 + m_{i}, ..., j + m_{i}, \phi_{i}(t)$, and $y_{i}(t)$.

When $j = n^*$, only the signal $\frac{s^{m_i+j}}{\Lambda_{ei}(s)}[u_i](t)$ needs to be considered. Concretely, $\frac{s^{m_i+j}}{\Lambda_{ei}(s)}[u_i](t) = \frac{s^{n_i}}{\Lambda_{ei}(s)}[u_i](t) = u_i(t) + \frac{s^{n_i}-\Lambda_{ei}(s)}{\Lambda_{ei}(s)}[u_i](t)$ with $\frac{s^{n_i}-\Lambda_{ei}(s)}{\Lambda_{ei}(s)}$ being strictly proper, which indicates the conclusion also holds for $j = n^*$. Thus, the lemma follows.

B.2. Proof of Lemma 4.2. We first demonstrate that $d_{i1}(t)$ converges to $s[y_i](t)$ by showing that the error term involving $\tilde{\theta}_{pi}(t)$ approaches zero as $t \to \infty$. Using mathematical induction, we extend this result to $d_{ik}(t)$, showing that it converges to $s^k[y_i](t)$ for higher orders. Combining these results, we then establish that the tracking error $\hat{r}_i(t) - r_i(t)$ converges to zero. The detailed proof process is as follows. With (3.8), we define

(B.5)
$$d_{ij}(t) = H_{ij}\left(y_i, \frac{s^{1+m_i}}{\Lambda_{ei}(s)}[u_i], \dots, \frac{s^{j+m_i}}{\Lambda_{ei}(s)}[u_i], \theta_{pi}, \phi_i\right),$$

for i = 1, ..., N and $j = 0, ..., n^*$. Comparing (3.8) and (B.5), we see that $d_{ij}(t)$, $j = 0, ..., n^*$, are the estimates of $y_i(t)$, $s[y_i](t), ..., s^{n^*}[y_i](t)$, respectively. Since $\dot{\theta}_{pi}(t) \in L^{\infty}$, $\dot{\omega}_{1i}^e(t) \in L^{\infty}$, $\dot{\omega}_{2i}^e(t) \in L^{\infty}$, $\dot{u}_i(t) \in L^{\infty}$, and $\dot{y}_i(t) \in L^{\infty}$, it follows that $\dot{r}_i(t) \in L^{\infty}$. Next, we will prove a stronger conclusion that

(B.6)
$$d_{ij}(t) - s^j[y_i](t) \to 0, \quad j = 0, \dots, n^*.$$

We now use mathematical induction to prove (B.6). The proving technique refers to the proof of the higher-order tracking property of MRAC in [38].

Let $\theta_{pi}(t) = \theta_{pi}(t) - \theta_{pi}^*$. When j = 1, from (B.1), the signal d_{i1} defined in (B.5) can be expressed by

(B.7)
$$d_{i1}(t) = \theta_{pi}^T(t)s[\phi_i](t) + \frac{s\Lambda_{i(n_i-1)}(s)}{\Lambda_{ei}(s)}[y_i](t).$$

Then, by (B.2) and (B.7), we have $d_{i1}(t) - s[y_i](t) = \tilde{\theta}_{pi}^T(t)s[\phi_i](t) = \tilde{\theta}_{pi}^T(t)\dot{\phi}_i(t)$. Noting (4.2) and (B.1), $\epsilon_i(t)$ can be expressed by $\epsilon_i(t) = \theta_{pi}^T(t)\phi_i(t) - \theta_{pi}^*T\phi_i(t) = \tilde{\theta}_{pi}^T(t)\phi_i(t)$. Then, the derivative of $\epsilon_i(t)$ is $\dot{\epsilon}_i(t) = \dot{\theta}_{pi}^T(t)\phi(t) + \tilde{\theta}_{pi}^T(t)\dot{\phi}_i(t)$. Noting (4.3), we have $\dot{\theta}_{pi}(t) \in L^\infty$ and thus $\dot{\epsilon}_i(t) \in L^\infty$. Hence, by (4.3), we have $\ddot{\theta}_{pi}(t) \in L^\infty$. Since $\dot{\theta}_{pi}(t) \in L^2$ by Lemma 4.1, then Lemma A.1 indicates that $\lim_{t\to\infty} \dot{\theta}_{pi}(t) = 0$. Thus, to prove that $d_{i1}(t) - s[y_i](t) = \tilde{\theta}_{pi}^T(t)\dot{\phi}_i(t)$ converges to zero, it is sufficient to prove $\lim_{t\to\infty} \dot{\epsilon}_i(t) = 0$. Next, we will prove this property by using the definition of limits, i.e., for any given η , there exists a $T = T(\eta) > 0$ such that $|\dot{\epsilon}_i(t)| < \eta$.

Copyright (c) by SIAM. Unauthorized reproduction of this article is prohibited.

(

We decompose the signal $\dot{\epsilon}_i(t)$ into two fictitious parts: one being small enough and one converging to zero asymptotically with time going to infinity. First, two fictitious K(s) and H(s) are introduced and defined by

(B.8)
$$K(s) = \frac{a^k}{(s+a)^k}, \quad sH(s) = 1 - K(s),$$

where a > 0 is an adjustable parameter. Thus, given K(s), the filter H(s) is strictly proper (with relative degree one) and stable, and is specified as

(B.9)
$$H(s) = \frac{1}{s}(1 - K(s)) = \frac{1}{s}\frac{(s+a)^k - a^k}{(s+a)^k}.$$

Moreover, from [28], it is known that the impulse response function of H(s) is $h(t) = \mathcal{L}^{-1}[H(s)] = e^{-at} \sum_{i=1}^{k} \frac{a^{k-i}}{(k-i)!} t^{k-i}$ and the L^1 signal norm of h(t) is

(B.10)
$$||h(\cdot)||_1 = \int_0^\infty |h(t)| dt = \frac{k}{a}.$$

We choose the filters K(s) and H(s) with k = 2. Using (B.8) that 1 = sH(s) + K(s), we divide $\dot{\epsilon}_i(t)$ into two terms:

$$\begin{aligned} \dot{\epsilon}_i(t) &= s[\tilde{\theta}_{pi}^T \phi_i](t) = H(s)s^2[\tilde{\theta}_{pi}^T \phi_i](t) + sK(s)[\tilde{\theta}_{pi}^T \phi_i](t) \\ &= H(s)s^2[\tilde{\theta}_{pi}^T \phi_i](t) + sK(s)[\epsilon_i](t). \end{aligned}$$
(B.11)

By the assumption $m_i(t) \in L^{\infty}$ and (B.3) and (B.4), we have $\phi_i(t), \dot{\phi}_i(t), \ddot{\phi}_i(t) \in L^{\infty}$. By Lemma 4.1, we have $\dot{\theta}_{pi}(t), \tilde{\theta}_{pi}(t) \in L^{\infty}$. Therefore, noting $\ddot{\theta}_{pi}(t) \in L^{\infty}$, it follows that

(B.12)
$$s^{2}[\tilde{\theta}_{pi}^{T}\phi_{i}](t) = [\ddot{\theta}_{pi}^{T}\phi_{i} + 2\dot{\theta}_{pi}^{T}\dot{\phi}_{i} + \tilde{\theta}_{pi}^{T}\ddot{\phi}_{i}](t) \in L^{\infty}$$

Then, from the above L^1 signal norm expression of H(s), $||h(\cdot)||_1 = \frac{2}{a}$, we have

(B.13)
$$\left| H(s)s^2[\tilde{\theta}_{pi}^T\phi_i](t) \right| \le \frac{c_1}{a}$$

for any $t \ge 0$ and some constant $c_1 > 0$ independent of a > 0. We now consider $sK(s)[\epsilon_i](t)$. Since $\dot{\phi}_i(t) \in L^{\infty}$ and $m_i(t) \in L^{\infty}$, then $\dot{\epsilon}_i(t) = \dot{\theta}_{pi}^T(t)\phi_i(t) + (\theta_{pi}(t) - \theta_{pi}^*)^T \dot{\phi}_i(t) \in L^{\infty}$. By Lemma 4.1 and $m_i(t) \in L^{\infty}$, we have $\epsilon_i(t) \in L^2$. Using Lemma A.1, it follows that $\lim_{t\to\infty} \epsilon_i(t) = 0$. Therefore, since sK(s) is stable and strictly proper, then, for any finite a > 0 in K(s),

(B.14)
$$\lim_{t \to \infty} sK(s)[\epsilon_i](t) = 0$$

For any $\eta > 0$, set $a = a(\eta) \ge \frac{2c_1}{\eta}$ for the filter H(s). Then, it follows that for any t > 0,

(B.15)
$$\left| H(s)s^2[\tilde{\theta}_{pi}^T\phi_i](t) \right| \le \frac{c_1}{a} \le \frac{\eta}{2}.$$

Moreover, by $\lim_{t\to\infty} sK(s)[\epsilon_i](t) = 0$, there exists $T = T(a(\eta), \eta) > 0$, such that for any t > T,

(B.16)
$$|sK(s)[\epsilon_i](t)| < \frac{\eta}{2}.$$

Therefore, due to (B.15) and (B.16), for any t > T,

(B.17)
$$|\dot{\epsilon}_i(t)| \le \left| H(s)s^2[\tilde{\theta}_{pi}^T\phi_i](t) \right| + |sK(s)[\epsilon_i](t)| < \frac{\eta}{2} + \frac{\eta}{2} = \eta,$$

which implies $\lim_{t\to\infty} \dot{\epsilon}_i(t) = 0$. So far we have proved that

$$\lim_{t \to \infty} \left(d_{i1}(t) - s[y_i](t) \right) = 0.$$

Given that for all $j = 1, ..., k - 1, k \leq n^*$, the following properties hold:

(B.18)
$$\lim_{t \to \infty} \epsilon_{i(k-1)}(t) = 0, \quad \lim_{t \to \infty} \left(d_{ij}(t) - s^j[y_i](t) \right) = 0,$$

where $\epsilon_{i(k-1)}(t) = \tilde{\theta}_{pi}^{T}(t) \left(s^{k-1}[\phi_i](t)\right)$. We have the following analysis.

When
$$j = k$$
, by (B.1), we have $s^k[y_i](t) = \theta_{pi}^{*T} s^k[\phi_i](t) + \frac{s^* \Lambda_{i(n_i-1)}(s)}{\Lambda_{ei}(s)}[y_i](t)$. Define

(B.19)
$$P(t) = s^{k}[\phi_{i}](t), Q(t) = \frac{s^{k}\Lambda_{i(n_{i}-1)}(s)}{\Lambda_{ei}(s)}[y_{i}](t).$$

Then,

(B.20)
$$s^k[y_i](t) = \theta_{pi}^{*T} P(t) + Q(t).$$

For simplicity of presentation, we denote

(B.21)
$$d_{ik}(t) = \theta_{pi}^T(t)\widehat{P}(t) + \widehat{Q}(t),$$

where $\hat{P}(t)$ and $\hat{Q}(t)$ are the estimates of P(t) and Q(t), respectively. Using (B.4), Q(t) and $\hat{Q}(t)$ can be expressed by

(B.22)
$$Q(t) = \sum_{l=0}^{k-1} \check{h}_l s^l [y_i](t) + \sum_{l=1}^{n_i-1} \check{l}_l \frac{s^l}{\Lambda_{ei}(s)} [y_i](t)$$

(B.23)
$$\widehat{Q}(t) = \sum_{l=0}^{k-1} \check{h}_l d_{il}(t) + \sum_{l=1}^{n_i-1} \check{l}_l \frac{s^l}{\Lambda_{ei}(s)} [y_i](t).$$

Then, by (B.22), (B.23), and the properties given in (B.18), we have

(B.24)
$$\lim_{t \to \infty} (\widehat{Q}(t) - Q(t)) = \lim_{t \to \infty} \left(\sum_{l=1}^{k-1} \widecheck{h}_l \left(d_{il} - s^l [y_i](t) \right) \right) = 0$$

Similarly, noting that each element of the vector $s^k[\phi_i](t)$ contains $s^{j-1}[y_i](t)$, $j = 1, \ldots, k$, and some filtered signals on $y_i(t)$ and $u_i(t)$, then by (B.4), (B.18), and similar analysis for the convergence of $\widehat{Q}(t) - Q(t)$, it follows that $\lim_{t\to\infty} (\widehat{P}(t) - P(t)) = 0$. Therefore, by (B.20) and (B.21), we have

(B.25)
$$\lim_{t \to \infty} (d_{ik}(t) - s^k[y_i](t)) = \lim_{t \to \infty} \tilde{\theta}_{pi}^T(t)P(t) + \lim_{t \to \infty} \theta_{pi}^T(t)(\hat{P}(t) - P(t)) + \lim_{t \to \infty} (\hat{Q}(t) - Q(t)) = \lim_{t \to \infty} \tilde{\theta}_{pi}^T(t)P(t).$$

We next prove that $\lim_{t\to\infty} \tilde{\theta}_{pi}^T(t)P(t) = \lim_{t\to\infty} \tilde{\theta}_{pi}^T(t) \left(s^k[\phi_i](t)\right) = 0$. Consider the signal $\epsilon_{i(k-1)}(t) = \tilde{\theta}_{pi}^T(t) \left(s^{k-1}[\phi_i](t)\right)$. Its derivative is

(B.26)
$$\dot{\epsilon}_{i(k-1)}(t) = \dot{\theta}_{pi}^T(t) s^{k-1}[\phi_i](t) + \tilde{\theta}_{pi}^T(t) s^k[\phi_i](t).$$

Copyright (C) by SIAM. Unauthorized reproduction of this article is prohibited.

Since $m_i(t) \in L^{\infty}$ and $\lim_{t\to\infty} \dot{\theta}_{pi}(t) = 0$, it follows that $\lim_{t\to\infty} \dot{\theta}_{pi}^T(t)s^{k-1}[\phi_i](t) = 0$. Hence, by (B.26), to prove $\lim_{t\to\infty} \tilde{\theta}_{pi}^T(t)(s^k[\phi_i](t)) = 0$, it is sufficient to prove $\lim_{t\to\infty} \dot{\epsilon}_{i(k-1)}(t) = 0$. Similar to (B.11), we express $\dot{\epsilon}_{i(k-1)}(t)$ as

(B.27)
$$\begin{aligned} \dot{\epsilon}_{i(k-1)}(t) &= s[\tilde{\theta}_{pi}^{T}\left(s^{k-1}[\phi_{i}]\right)](t) \\ &= H(s)s^{2}[\tilde{\theta}_{pi}^{T}\left(s^{k-1}[\phi_{i}]\right)](t) + sK(s)[\epsilon_{i(k-1)}](t). \end{aligned}$$

By the assumption $m_i(t) \in L^{\infty}$ and (B.3) and (B.4), we have, for $k \leq n^*$, $s^k \phi_i(t) \in L^{\infty}$. When $k = n^*$, by the additional assumption $\dot{u}_i(t), \dot{y}_i(t) \in L^{\infty}$, we have $s^{k+1}\phi_i(t) \in L^{\infty}$. Moreover, by Lemma 4.1, we have $\dot{\theta}_{pi}(t), \tilde{\theta}_{pi}(t) \in L^{\infty}$. Therefore, noting $\ddot{\theta}_{pi}(t) \in L^{\infty}$, it follows that

$$s^{2}[\tilde{\theta}_{pi}^{T}(t)\left(s^{k-1}[\phi_{i}]\right)](t) = \left[\ddot{\theta}_{pi}^{T}s^{k-1}[\phi_{i}] + 2\dot{\theta}_{pi}^{T}s^{k}[\phi_{i}] + \tilde{\theta}_{pi}^{T}s^{k+1}[\phi_{i}]\right](t) \in L^{\infty}.$$

Then, for j = k, similar to (B.13), we have $|H(s)s^2[\tilde{\theta}_{pi}^T s^{k-1}[\phi_i]](t)| \leq \frac{c_k}{a}$, for some $c_k > 0$ independent of a. Since sK(s) is stable and strictly proper, so that, with $\lim_{t\to\infty} \epsilon_{i(k-1)}(t) = 0$, we have $\lim_{t\to\infty} sK(s)[\epsilon_{i(k-1)}](t) = 0$. Hence, similar to (B.17), by choosing suitable parameter a > 0 in H(s) and K(s), it can be shown that for any $\eta > 0$, there exists $T = T(\eta, a) > 0$, such that for any t > T, it holds that $|\dot{\epsilon}_{i(k-1)}(t)| < \eta$. Therefore, $\lim_{t\to\infty} \dot{\epsilon}_{i(k-1)}(t) = 0$. Then, by $\lim_{t\to\infty} \dot{\theta}_{pi}^T(t)s^{k-1}[\phi_i](t) = 0$ as established above (B.27), and (B.25), we have

(B.28)
$$\lim_{t \to \infty} \epsilon_{ik}(t) = \lim_{t \to \infty} \tilde{\theta}_{pi}^T(t) \left(s^k[\phi_i](t) \right) = 0, \quad \lim_{t \to \infty} \left(d_{ik}(t) - s^k[y_i](t) \right) = 0.$$

Therefore, by (3.8), (3.9), (4.9), and (B.5), it follows that

$$\hat{r}_i(t) - r_i(t) = \sum_{j=0}^{n^*} \psi_j \left(d_{ij}(t) - s^j [y_i](t) \right) \to 0,$$

with ψ_i defined below (3.9). The proof is completed.

REFERENCES

 \Box

- M. BÖHM, M. A. DEMETRIOU, S. REICH, AND I. G. ROSEN, Model reference adaptive control of distributed parameter systems, SIAM J. Control Optim., 36 (1998), pp. 33–81, https://doi.org/10.1137/S0363012995279717.
- [2] H. CAI, F. L. LEWIS, G. HU, AND J. HUANG, The adaptive distributed observer approach to the cooperative output regulation of linear multi-agent systems, Automatica, 75 (2017), pp. 299–305, https://doi.org/10.1016/j.automatica.2016.09.038.
- [3] P. CHVYKOV, T. A. BERRUETA, A. VARDHAN, W. SAVOIE, A. SAMLAND, T. D. MUR-PHEY, K. WIESENFELD, D. I. GOLDMAN, AND J. L. ENGLAND, Low rattling: A predictive principle for self-organization in active collectives, Science, 371 (2021), pp. 90–95, https://doi.org/10.1126/science.abc6182.
- [4] J. A. FAX AND R. M. MURRAY, Information flow and cooperative control of vehicle formations, IEEE Trans. Automat. Control, 49 (2004), pp. 1465–1476, https://doi.org/ 10.1109/TAC.2004.834433.
- [5] R. GOEL, T. GARG, AND S. B. ROY, Closed-loop reference model based distributed MRAC using cooperative initial excitation and distributed reference input estimation, IEEE Trans. Control Network Syst., 9 (2022), pp. 37–49, https://doi.org/10.1109/TCNS.2022.3141015.
- [6] G. GUO, J. KANG, R. R. LI, AND G. H. YANG, Distributed model reference adaptive optimization of disturbed multiagent systems with intermittent communications, IEEE Trans. Cybernetics, 52 (2020), pp. 5464–5473, https://doi.org/10.1109/TCYB.2020.3032429.
- [7] H. HAGHSHENAS, M. A. BADAMCHIZADEH, AND M. BARADARANNIA, Containment control of heterogeneous linear multi-agent systems, Automatica, 54 (2015), pp. 210–216, https://doi.org/10.1016/j.automatica.2015.02.002.

- [8] T. R. HAN, Z. Y. LIN, R. H. ZHENG, AND M. Y. FU, A barycentric coordinate-based approach to formation control under directed and switching sensing graphs, IEEE Trans. Cybernetics, 48 (2017), pp. 1202–1215, https://doi.org/10.1109/TCYB.2017.2684461.
- [9] Y. G. HONG, G. R. CHEN, AND L. BUSHNELL, Distributed observers design for leaderfollowing control of multi-agent networks, Automatica, 44 (2008), pp. 846–850, https://doi.org/10.1016/j.automatica.2007.07.004.
- [10] P. A. IOANNOU AND J. SUN, Robust Adaptive Control, Courier Corporation, 2012.
- [11] A. JADBABAIE, J. LIN, AND A. S. MORSE, Coordination of groups of mobile autonomous agents using nearest neighbor rules, IEEE Trans. Automat. Control, 48 (2003), pp. 988–1001, https://doi.org/10.1109/TAC.2003.812781.
- [12] H. K. KHALIL AND J. W. GRIZZLE, Nonlinear Systems, 3rd ed., Prentice Hall, Upper Saddle River, NJ, 2002, https://testbankdeal.com/sample/nonlinear-systems-3rd-edition-khalilsolutions-manual.pdf.
- [13] M. KRSTIC, P. V. KOKOTOVIC, AND I. KANELLAKOPOULOS, Nonlinear and Adaptive Control Design, John Wiley & Sons, 1995.
- [14] S. G. LI, R. BATRA, D. BROWN, H.-D. CHANG, N. RANGANATHAN, C. HOBERMAN, D. RUS, AND H. LIPSON, Particle robotics based on statistical mechanics of loosely coupled components, Nature, 567 (2019), pp. 361–365, https://doi.org/10.1038/s41586-019-1022-9.
- [15] Y. LI, X. WANG, J. SUN, G. WANG, AND J. CHEN, Data-driven consensus control of fully distributed event-triggered multi-agent systems, Sci. China Inf. Sci., 66 (2023), 152202, https://doi.org/10.1007/s11432-022-3629-1.
- [16] Z. K. LI AND Z. T. DING, Distributed adaptive consensus and output tracking of unknown linear systems on directed graphs, Automatica, 55 (2015), pp. 12–18, https://doi.org/ 10.1016/j.automatica.2015.02.033.
- [17] Z. K. LI, Z. S. DUAN, AND G. R. CHEN, Dynamic consensus of linear multi-agent systems, IET Control Theory Appl., 5 (2011), pp. 19–28, https://doi.org/10.1049/iet-cta.2009.0466.
- [18] Z. Y. LIN, L. L. WANG, Z. M. HAN, AND M. Y. FU, Distributed formation control of multiagent systems using complex Laplacian, IEEE Trans. Automat. Control, 59 (2014), pp. 1765–1777, https://doi.org/10.1109/TAC.2014.2309031.
- [19] H. Y. LIU, G. M. XIE, AND L. WANG, Necessary and sufficient conditions for containment control of networked multi-agent systems, Automatica, 48 (2012), pp. 1415–1422, https://doi.org/10.1016/j.automatica.2012.05.010.
- [20] M. LU AND L. LIU, A low gain approach to output consensus of networked heterogeneous linear multi-agent systems, SIAM J. Control Optim., 59 (2021), pp. 4295–4313, https://doi.org/10.1137/19M1296458.
- [21] D. G. LUI, A. PETRILLO, AND S. SANTINI, Distributed model reference adaptive containment control of heterogeneous multi-agent systems with unknown uncertainties and directed topologies, J. Frankl. Inst., 358 (2021), pp. 737–756, https://doi.org/ 10.1016/j.jfranklin.2020.11.005.
- [22] Z. MENG, G. SHI, AND K. H. JOHANSSON, Multiagent systems with compasses, SIAM J. Control Optim., 53 (2015), pp. 3057–3080, https://doi.org/10.1137/140982283.
- [23] E. NUNO, R. ORTEGA, L. BASANEZ, AND D. HILL, Synchronization of networks of nonidentical Euler-Lagrange systems with uncertain parameters and communication delays, IEEE Trans. Automat. Control, 56 (2011), pp. 935–941, https://doi.org/ 10.1109/TAC.2010.2103415.
- [24] R. OLFATI-SABER AND R. M. MURRAY, Consensus problems in networks of agents with switching topology and time-delays, IEEE Trans. Automat. Control, 49 (2004), pp. 1520–1533, https://doi.org/10.1109/TAC.2004.834113.
- [25] Z. H. PENG, D. WANG, H. W. ZHANG, G. SUN, AND H. WANG, Distributed model reference adaptive control for cooperative tracking of uncertain dynamical multi-agent systems, IET Control Theory Appl., 7 (2013), pp. 1079–1087, https://doi.org/10.1049/iet-cta.2012.0765.
- [26] W. REN, Synchronization of coupled harmonic oscillators with local interaction, Automatica, 44 (2008), pp. 3195–3200, https://doi.org/10.1016/j.automatica.2008.05.027.
- [27] A. SARLETTE, R. SEPULCHRE, AND N. E. LEONARD, Autonomous rigid body attitude synchronization, Automatica, 45 (2009), pp. 572–577, https://doi.org/10.1016/ j.automatica.2008.09.020.
- [28] S. SASTRY, M. BODSON, AND J. F. BARTRAM, Adaptive Control: Stability, Convergence, and Robustness, Prentice-Hall, Upper Saddle River, NJ, 1990.
- [29] G. SONG, Adaptive Control for Distributed Leader-Following Consensus of Multi-Agent Systems, Ph.D. thesis, University of Virginia, Charlottesville, VA, 2015.
- [30] G. SONG AND G. TAO, An adaptive consensus control scheme for multi-agent systems with persistent switching topology, in Proceedings of the 2018 American Control Conference, IEEE, 2018, pp. 2509–2514.

- [31] Q. SONG, J. D. CAO, AND W. W. YU, Second-order leader-following consensus of nonlinear multi-agent systems via pinning control, Systems Control Lett., 59 (2010), pp. 553–562, https://doi.org/10.1016/j.sysconle.2010.06.016.
- [32] Y. F. SU AND J. HUANG, Cooperative output regulation of linear multi-agent systems, IEEE Trans. Automat. Control, 57 (2011), pp. 1062–1066, https://ieeexplore. ieee.org/abstract/document/6026912.
- [33] Y. F. SU AND J. HUANG, Cooperative output regulation with application to multi-agent consensus under switching network, IEEE Trans. Syst. Man Cybernet. Part B (Cybernetics), 42 (2012), pp. 864–875, https://doi.org/10.1109/TSMCB.2011.2179981.
- [34] Y. F. SU AND J. HUANG, Two consensus problems for discrete-time multi-agent systems with switching network topology, Automatica, 48 (2012), pp. 1988–1997, https://doi.org/ 10.1016/j.automatica.2012.03.029.
- [35] Y. F. SU AND J. HUANG, Cooperative global output regulation of heterogeneous secondorder nonlinear uncertain multi-agent systems, Automatica, 49 (2013), pp. 3345–3350, https://doi.org/10.1016/j.automatica.2013.08.001.
- [36] H. G. TANNER, G. J. PAPPAS, AND V. KUMAR, Leader-to-formation stability, IEEE Trans. Robot. Automat., 20 (2004), pp. 443–455, https://doi.org/10.1109/TRA.2004.825275.
- [37] G. TAO, Adaptive Control Design and Analysis, Vol. 37, John Wiley & Sons, 2003.
- [38] G. TAO AND G. SONG, Higher order tracking properties of model reference adaptive control systems, IEEE Trans. Automat. Control, 63 (2018), pp. 3912–3918, https://doi.org/ 10.1109/TAC.2018.2802496.
- [39] M. E. VALCHER AND P. MISRA, On the consensus and bipartite consensus in high-order multiagent dynamical systems with antagonistic interactions, Systems Control Lett., 66 (2014), pp. 94–103, https://doi.org/10.1016/j.sysconle.2014.01.006.
- [40] M. WANG, Z. D. WANG, H. L. DONG, AND Q. L. HAN, A novel framework for backstepping-based control of discrete-time strict-feedback nonlinear systems with multiplicative noises, IEEE Trans. Automat. Control, 66 (2020), pp. 1484–1496, https://doi.org/ 10.1109/TAC.2020.2995576.
- [41] X. L. WANG, Y. G. HONG, J. HUANG, AND Z. P. JIANG, A distributed control approach to a robust output regulation problem for multi-agent linear systems, IEEE Trans. Automat. Control, 55 (2010), pp. 2891–2895, https://doi.org/10.1109/TAC.2010.2076250.
- [42] L. WEN, G. TAO, B. JIANG, AND H. YANG, Adaptive LQ control using reduced Hamiltonian for continuous-time systems with unmatched input disturbances, SIAM J. Control Optim., 59 (2021), pp. 3625–3660, https://doi.org/10.1137/19M1275115.
- [43] Y. WU, Q. LIANG, Y. ZHAO, J. HU, AND L. XIANG, Distributed estimation-based output consensus control of heterogeneous leader-follower systems with antagonistic interactions, Sci. China Inf. Sci., 66 (2023), 139204, https://doi.org/10.1007/s11432-020-3191-x.
- [44] Y. K. XIE AND Q. MA, Consensus analysis of fractional multi-agent systems with delayed distributed PI controller, J. Syst. Sci. Complex., 36 (2023), pp. 205–221, https://doi.org/10.1007/s11424-022-1256-8.
- [45] Y. C. XU, Y. J. ZHANG, AND J. F. ZHANG, Singularity-free adaptive control of MIMO DT nonlinear systems with general vector relative degrees, Automatica, 153 (2023), 111054, https://doi.org/10.1016/j.automatica.2023.111054.
- [46] K. Y. YOU AND L. H. XIE, Network topology and communication data rate for consensusability of discrete-time multi-agent systems, IEEE Trans. Automat. Control, 56 (2011), pp. 2262– 2275, https://doi.org/10.1109/TAC.2011.2164017.
- [47] C. Z. YUAN, W. ZENG, AND S. L. DAI, Distributed model reference adaptive containment control of heterogeneous uncertain multi-agent systems, ISA Trans., 86 (2019), pp. 73–86, https://doi.org/10.1016/j.isatra.2018.11.003.
- [48] Y. J. ZHANG AND J. F. ZHANG, A quantized output feedback MRAC scheme for discrete-time linear systems, Automatica, 145 (2022), 110575, https://doi.org/10.1016/ j.automatica.2022.110575.
- [49] Y. J. ZHANG, J. F. ZHANG, AND X. K. LIU, Matrix decomposition-based adaptive control of noncanonical form MIMO DT nonlinear systems, IEEE Trans. Automat. Control, 67 (2021), pp. 4330–4337, https://doi.org/10.1109/TAC.2021.3115850.
- [50] Z. Y. ZHEN, G. TAO, Y. XU, AND G. SONG, Multivariable adaptive control based consensus flight control system for UAVs formation, Aerosp. Sci. Technol., 93 (2019), 105336, https://doi.org/10.1016/j.ast.2019.105336.
- [51] X. ZONG, T. LI, AND J. F. ZHANG, Consensus conditions of continuous-time multi-agent systems with additive and multiplicative measurement noises, SIAM J. Control Optim., 56 (2018), pp. 19–52, https://doi.org/10.1137/15M1019775.